CREB Working Paper No. 03-12

The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

Rabia Arif, Azam Chaudhry



Centre for Research in Economics and Business Lahore School of Economics

Centre for Research in Economics and Business (CREB)

Naved Hamid Director CREB

CREB Advisory Board

Shahid Amjad Chaudhry Rector Lahore School of Economics

Sohail Zafar Dean Faculty of Business Administration **Azam Chaudhry** Dean Faculty of Economics Muneer Ahmed Director Centre for Policy and Environmental Studies

Shahid Siddiqui Director Centre for Humanities and Social Sciences Rana Wajid Director Centre for Mathematics and Statistical Sciences **Iqbal M. Khan** Editor Lahore School Case Study Journal

Lahore School of Economics

Intersection Main Boulevard Phase VI, DHA and Burki Road Lahore 53200, Pakistan Tel: 042-36561230; 042-36560936 Email: creb@lahoreschool.edu.pk

The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

Rabia Arif

Assistant Professor Lahore School of Economics

Azam Chaudhry

Associate Professor Lahore School of Economics © 2012 Centre for Research in Economics and Business Lahore School of Economics All rights reserved.

First printing July 2012.

The views expressed in this document are those of the authors and do not necessarily reflect the views of the Centre for Research in Economics and Business or the Lahore School of Economics.

Lahore School of Economics Intersection of Main Boulevard, Phase VI, DHA and Burki Road Lahore 53200, Pakistan Tel.: +92 42 3656 1230 creb@lahoreschool.edu.pk www.creb.org.pk

Price: Rs100

Preface

The Centre for Research in Economics and Business (CREB) was established in 2007 to conduct policy-oriented research with a rigorous academic perspective on key development issues facing Pakistan. In addition, CREB (i) facilitates and coordinates research by faculty at the Lahore School of Economics, (ii) hosts visiting international scholars undertaking research on Pakistan, and (iii) administers the Lahore School's postgraduate program leading to the MPhil and PhD degrees.

An important goal of CREB is to promote public debate on policy issues through conferences, seminars, and publications. In this connection, CREB organizes the Lahore School's Annual Conference on the Management of the Pakistan Economy, the proceedings of which are published in a special issue of the *Lahore Journal of Economics*.

The CREB Working Paper Series was initiated in 2008 to bring to a wider audience the research being carried out at the Centre. It is hoped that these papers will promote discussion on the subject and contribute to a better understanding of economic and business processes and development issues in Pakistan. Comments and feedback on these papers are welcome.

Abstract

In developing countries, external migration tends to ease household income constraints because of the external remittances that are generated. Using data on the Punjab, this working paper attempts to determine whether the external migration of individuals in a household has a positive effect on children's schooling outcomes as measured by school enrolments, the accumulated level of schooling, number of days spent in school, and dropouts. We use historic migration rates to instrument for migration in an analysis of school outcomes for children aged 5–17, 5–11, and 12–17 to determine which group is most affected by external migration. The results show that external migration has a significantly positive impact on the school enrolments of younger children, whereas the accumulated level of schooling among older children increases significantly if there is an external migrati in their households.

Keywords: Migration, schooling, enrolments, Pakistan.

JEL classifications: l21, C26.

The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

1. Introduction

Despite the obvious benefits of education for socioeconomic development, the United Nation's statistics reveal that barely 63 percent of Pakistani children complete primary school education, while only 68 percent of Pakistani boys and 72 percent of Pakistani girls reach Grade 5 (United Nations Population Fund, 2009). One of the major reasons that Pakistan, like other developing countries, suffers from such a low level of private investment in education is due to liquidity constraints, which, in turn, can be the result of incomplete or missing credit markets (Jacoby, 1994; Jacoby & Skoufias, 1997). Therefore, remittances from migrants play a crucial role in easing, if not completely eliminating, such constraints.

Even though individuals migrate to improve their own living standards, a commonly asked question is how their migration affects their family members back home. Over the last 50 years, a significant number of people have emigrated from Pakistan, creating a vast network of migrants around the world, which, in turn, has reduced the formal and informal costs associated with migration. These networks reduce formal costs by sending back the remittances earned and informal costs by decreasing the barriers of asymmetric information regarding the opportunities in developed countries for labor in developing countries. According to international statistics, Pakistan falls within the top ten remittance-recipient countries (World Bank, 2006) and is also among the top ten immigration rate has always been negative, indicating that the emigration of individuals from Pakistan has always outweighed the immigration of individuals to Pakistan.

Data from the Bureau of Emigration and Overseas Migration suggests that 60 percent of the total number of people who migrated over time came from only 20 districts of Pakistan—mainly in northern Punjab, two districts from southern Punjab, Khyber Pakhtunkhwa, and Sindh (Arif, 2009) The historic migration rates for Punjab, calculated as the total number of people who migrated abroad divided by the total population of Punjab over time, are shown in Figure 1.



Figure 1: External migration rates for Punjab

Source: Government of Pakistan, Bureau of Emigration and Overseas Employment (2009); Government of Pakistan, Population Census Organization (1998).

Figure 1 shows that there have been fluctuations in the migration rate since 1981, and that it rose substantially in 2007–09 and declined drastically afterward.

The aim of this working paper is to explore the relationship between external migration and school enrolments in Pakistan. This will contribute to the debate on whether migration promotes human development in Pakistan through the stream of cash inflows generated via remittances, or whether remittances are spent merely on the consumption of goods and services together with households' leisure consumption, allowing us to understand which of the two effects dominate in the case of the Punjab. Our approach contributes to the existing literature by using historic migration rates in the Punjab as an instrument for an external migrant dummy.

The paper is structured as follows: This introduction is followed by a review of the literature. Section 3 puts forward a basic model of schooling. Section 4 presents an econometric model of schooling that incorporates external migration, together with the limitations of the specifications, the sources of the data used, and summary statistics.

Section 5 discusses the results of our estimations of the impact of historic migration rates on current migration. Section 6 presents the results of our estimations of the impact of external migration on enrolments, accumulated level of schooling, and days spent by a child in school last week. Section 7 reports the results of robustness checks, and Section 8 concludes the study.

2. A Review of the Literature

A large body of research has looked at the relationship between migration and remittances. Broadly speaking, this literature can be divided into three general categories: (i) studies that identify the determinants of migration, (ii) studies that examine the effects of migration on the overall development of countries, and (iii) those that review the effects of migration on educational attainment. Our paper is an extension of the last category and focuses on how external migration affects school enrolments and other educational outcomes.

Many analyses of migration have looked at factors affecting the decision to migrate. Authors such as Mincer (1978) and Hoddinott (1994) take migration as a dependent variable, present models that give the possible determinants of migration, and discuss how an individual's decision to migrate depends not only on his/her utility function but also on the family's utility function. Therefore, the decision to migrate is an outcome of both the individual's and household's utility function (Hoddinot, 1994; Rapoport & Docquier, 2005; Stark & Lucas, 1988). According to Hoddinott (1994), the level of remittances is influenced by a parent's ability to reward good behavior through the promise of bequests. In poor countries, migration is financed mostly by contributions from the resources of other family members. Thus, the migrant compensates for this investment by remitting to his/her household or family members. This argument is supported by the fact that the migrating member of the family usually remits to his/her household or family members.

Another branch of the literature looks at how migration eases constraints and improves opportunities in migrants' home countries. This is brought about by enhanced standards of living and is achieved mainly by reducing poverty, increasing employment and health expenditures, and bringing about improvements in education directly and indirectly via remittances. Macroeconomic analyses by Russell (1992), Taylor, Arango, Hugo, Kouaouci, Massey, and Pellegrino (1996a, 1996b), Massey and Basem (1992), and Martin (1990) across countries indicate that these remittances are used for daily expenses such as food and clothing, and that they compose a significant portion of these households' income. Instead of being spent on education, healthcare, and small businesses, remittances are spent on building or improving housing, and buying land, cattle, or consumer goods.

Using a simple ordinary least squares (OLS) analysis, Adams (1998) studies the relationship between remittances and rural asset accumulation in Pakistan, and concludes that these households are less inclined to sacrifice their present consumption. Other studies, however, have found evidence supporting the argument that the remittances collected by a migrant's household are used consciously in productive investments since they are taken as an uncertain stream of income. Studies carried out by Adams (2005), Conway and Cohen (1998), and Kugler (2006) indicate that (i) females in the migrant's household have more influence in making decisions about capital inflows, (ii) villagers gain more experience by working outside, and (iii) overall, migration is the "lesser evil" despite the fact that family life may be disrupted by an absent family member. Essentially, migration creates high labor demand in the home market, thus increasing job opportunities for people who have not migrated.

Since this working paper analyzes the effect of migration on primary school enrolment in Punjab, it is important to link the decision to migrate with those concerning enrolments and schooling. The classical model of human capital investments over the life cycle formulated by Ben-Porath (1967) shows how the benefits of schooling change over time. The model suggests that the production function of human capital is concave and that the opportunity cost of allocating time to further skill acquisition increases as skills accumulate over time. Therefore, the marginal returns associated with schooling diminish over time. Moreover, a finite life span limits the time period available to capture returns from schooling as age increases. Hence, as age increases, the marginal returns to time in school tend to decrease after a specific age.

Becker (1991) has similarly argued that if, early in life, the present value of the return is sufficiently high relative to its current marginal cost, children will choose to study. Using a household production framework,

Becker and Tomes (1976) build on Ben-Porath (1967) and discuss how the educational decision-making process is important to household welfare. According to them, parents are altruistic in nature and their utility depends on that of the entire family. Parents' primary concern, however, is to maximize family wealth and, hence, to invest more in children if these children have higher academic potential and are likely to earn higher wages in future to support their parents. Likewise, the amount that poorer parents invest in their children's education will be smaller than the optimal, but the amount being invested will rise with an increase in the level of parents' income.

Another theory put forward by Sawada and Lokshin (2001) discusses how parents may end up choosing "winners" and allocate more resources toward their education. As an initial theoretical framework to account for this aspect of household behavior, the authors employ two sets of optimal behavioral rules. First, for the inter-temporal allocation of resources, parents decide to maximize the expected total lifetime utility of their family. Second, given an overall resource constraint, parents choose an optimal allocation of educational resources among their children.

Building on the theoretical analyses above, researchers have also empirically estimated the effect of external migration and remittances on enrolments different in countries. Acosta (2006)presents microeconomic evidence on the economic effects of remittances on household spending decisions. Average estimates suggest that young girls and boys (under 15 years of age) from remittance-recipient households are more likely to be enrolled in school. He concludes that remittances can affect household labor supply and investment in children's human capital. The study's results also suggest that, when differentiated on the basis of demographic groups, young girls and boys-11-14 years old-are seen to benefit from remittances in terms of higher enrolment rates, but that this positive impact does not apply to older—15–17-year-old—boys.

Remittances are also found to act as a substitute for child labor, resulting in higher school dropout rates. Mansuri (2006), Edwards and Ureta (2003), Hanson and Woodruff (2003), and Alcaraz, Chiquiar, and Salcedo (2012) have conducted similar studies for different countries and concluded that there is a positive relationship between primary enrolments and migration. Children receiving remittances with parents who had attained low levels of schooling showed low levels of retention. Children in households with migrants were able to complete more years of schooling, although their parents had low levels of education. Empirical estimates also show that the remittances crisis has had a negative and significant effect on children's schooling and a positive and significant effect on child labor.

Our analysis is relatively unique in that we have used historic migration rates as an instrument in our analysis of the impact of migration on school enrolment, as well as an improved set of explanatory variables to account for the variations in schooling outcomes.

3. A Basic Model of Schooling

The theory we use builds on that used by Sawada and Lokshin (2001), who employ a standard investment model of education as a benchmark and apply it to Pakistan. Their model draws on work by Levhari and Weiss (1974) and Jacoby and Skoufias (1997) on human capital investment under uncertainty. They extend the Jacoby and Skoufias model to a generalized form with multiple children. Essentially, risk, uncertainty, and constraints to insurance and credit influence poor Pakistani families' investment and consumption decisions. Therefore, Sawada and Lokshin formalize human capital accumulation in rural Pakistan as households' sequential schooling investment decisions under uncertainty and credit constraints. The model is derived for Pakistan and further adapted by adding to it, additional sets of variables.

As per Sawada and Lokshin's (2001) model, let us assume that a household with *n* children chooses household consumption *C* and schooling for child *i*, S_i , so as to maximize the household's aggregated expected utility with a concave instantaneous utility function, $U(\bullet)$, given the information set at the beginning of time t, Ω_t . The information set Ω_t includes initial asset ownership and the entire history of household variables. Therefore, the household's problem can be represented as

$$MaxE_{\{C_{t},S_{it}\}}\left\{\sum_{k=0}^{T-t}\beta^{k}U(C_{t+k})+\beta^{T+1}W(A_{T+1},H_{1T+1}^{C},H_{2T+1}^{C},...,H_{nT+1}^{C})\right|\Omega_{t}\right\}$$
(1)

where $B \ge 0$, and H^p , A_0 , and B_0 are given. In this problem, the objective function includes a concave function, $W(\bullet)$, which comprises a financial bequest and the salvage value of the final stock of the child's

human capital. The parameter β represents the discount factor. The above household utility function is then maximized, given the following constraints:

$$A_{t+1} = A_t + Y_t(H^p) + \sum_{i=1}^n W(1 - S_{it}) - C_t(1 + r_t)$$
(2)

$$H_{it+1}^{C} = H_{it}^{C} + \sum_{i=1}^{n} [f(S_{it}, q_{it}) + e_{it}] \text{ where } i = 1, 2, ..., n$$
(3)

$$A_{t} + Y_{t}(H^{p}) + \sum_{i=1}^{n} Wi(1 - S_{it}) + B \ge C_{t}$$
(4)

The first constraint, equation (2), is the household's inter-temporal budget constraint. The household's consumable resources in each period comprise assets, A, where $A_T \ge 0$; stochastic parental income, Y, which is a function of parents' human capital H_p ; and total child income, $\Sigma W_i(1 - S_{ii})$, with W_i being the child-specific wage rate.

The second constraint, equation (3), is the human capital accumulation equation. The human capital production function, $f(\bullet)$, includes the variable q, which represents the school supply-side effect, the gender gap, and subjective factors. Among others, the variable q is a function of a time-invariant gender dummy variable that takes the value of 1 if the child is female and 0 if the child is male. Lastly, there is an additive stochastic element, e, which incorporates possibilities of exogenous shocks such as the risk of job mismatching after schooling and poor market conditions, etc. We assume that e is a stochastic error term, therefore $(e_{ir}|\Omega_r) = 0$ for all i.

The third constraint—equation (4)—represents the potentially binding credit constraint where *B* is the maximum amount of credit available to the household. According to Sawada and Lokshin (2001), this stochastic programming model has n+1 state variables: physical assets, *A*, and child human assets, H_i^c , i = 1, 2, ..., n. When income is stochastic, analytical solutions to this problem—even without human capital—cannot be derived in general (Zeldes, 1989). The standard first-order conditions are derived to calculate the optimum conditions. Hence, the Kuhn-Tucker conditions are applied to the standard Bellman equation. The arguments below use the first-order conditions of the above problem.

Under credit market imperfections, the household's shadow interest rate is determined endogenously; it is given by the marginal rate of substitution of consumption over time because the separability between consumption and schooling investment decisions breaks down as indicated by equation (5) below. This implies that, once the constraint becomes binding—that is, once households are unable to save and borrow money freely—then schooling decisions over time are not independent of consumption decisions. The equalization of the marginal rate of transformation with the marginal rate of substitution would give the optimal condition, which is

$$\frac{\frac{\partial f}{\partial S_{it}}}{\frac{\partial f}{\partial S_{it} - 1}} = \beta E_{t-1} \left[\frac{\partial U}{\partial C_t} / \frac{\partial U}{\partial C_{t-1}} \right], \forall i$$
(5)

$$S_{it}^* = X_{it}\beta^C + S_{-it}^* + \varepsilon_{it}, \forall i$$
(6)

According to equation (6) above, the optimum decision for child i is conditional on the decision made for all other children in that household. The optimal choice for child *i*'s schooling, S_i^* , depends on S_{-i}^* , where S_i^* is the schooling decision made for child I and S^* is the optimal schooling decision made for a child other than *i*. X includes the gender indicator variable. the access-to-school variable, and household-specific characteristics such as educational investments and the ownership and accumulation of human and physical assets, etc. In contrast to a household with perfect credit availability, where the parents' income does not affect the child's schooling, a household facing credit constraints would associate high marginal costs with its children's schooling if it were to experience a negative income shock. This reflects that consumption and schooling decisions are not separable under a binding credit constraint.

Based on the model above, one can conclude that, in developing countries such as Pakistan, where the limited-income constraint is the most important concern of individuals in their everyday life, decisions such as that of their children's schooling are indeed influenced by household income. Additionally, other factors such as children's own characteristics, household characteristics, and geographical and demographic indicators play an important role in determining their schooling.

4. A Model for Estimating the Impact of External Migration on Enrolments

4.1. A Schooling Model Incorporating External Migration

In order to derive an econometric model for the schooling of a child in Pakistan, we start with a simplified version of equation (6) (Section 3):

$$S_{it}^* = X_{it}\beta_t + \mu_{it} \tag{7}$$

where S_{it}^* is a dummy dependent variable that takes the value of 1 if the child attends school and 0 otherwise; and X_{it} includes the gender indicator variable, school supply variables, determinants of the household's preferences, household shock variables, and sibling composition. Therefore, unlike other economic decisions, the schooling decision is not based on a simple cost–benefit analysis (see Acosta, 2006; Edwards & Ureta, 2003; Hanson & Woodruff, 2003; Mansuri, 2006). Both household and geographical characteristics—the distance from the house to the school and whether the household lives in a remote area—together with the child's own characteristics are important in determining whether or not that child will be enrolled in school, especially in a developing country (see Edwards & Ureta, 2003; Hanson & Woodruff, 2003).

Since the dataset being used in this case is cross-sectional, so is the variation in schooling enrolment that will be incorporated in the model. Children with different characteristics such as age, gender, and ability will have different outcomes for enrolment. Even a simple cost–benefit analysis indicates that parents associate higher benefits with children who show higher ability. An unobserved characteristic could, however, be partially observed by incorporating parents' education in the model since parents' education is a correlate of children's ability—educated parents are more likely to have higher-ability children (as an inherited characteristic). In the context of developing countries, gender disparity is also a major concern, and a child's gender plays a vital role in his/her schooling decision process. Consequently, we also add a gender dummy to the model.

The bulk of the literature refers to school enrolment as a function of household characteristics. These include observed characteristics such as household income and level of education, but characteristics such as parents' ability are only partially observed, and others, such as parents' preferences, are unobserved. We therefore add all observed and partially observed household characteristics such as parents' education, age, and status to the final model. The number of siblings and their respective ages, which act as a resource constraint to parents, also plays an important role in the decision making process and are included in the model.

In addition to the variables discussed in equation (7), there are several ways in which household composition can affect the child's enrolment. Disruptions to family life in single-parent- or divorced-parent-households may affect the resources they devote to educating their children (Hanson & Woodruff, 2003). We therefore add several variables to explain these disruptions, such as the presence of a female household head or whether or not the parents are divorced.

Based on the discussion above, we will estimate the following equation:

$$S_{ghi} = \beta_1 M_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{ghi} + \mu_{ghi}$$
(8)

Equation (8) incorporates those factors that affect the measure of schooling attained by child g from a given household residing in a given district. S_{ghi} is the measure of the child's schooling, and is equal to 1 if child g in household h and district i is enrolled and 0 otherwise. C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, and B_{hi} is the child's gender. M_{ghi} is introduced to incorporate the effect of migration; this variable takes the value of 1 if the household includes a migrant and 0 otherwise.

A potential problem with this specification is that of endogeneity. Since unobserved household and district characteristics simultaneously influence both the decision to migrate and schooling decision, any OLS estimates would be biased. So, migration in this specification is an endogenous variable and an instrumental variable is required to obtain unbiased estimates.

Another point worth noting is that, in this empirical analysis, the migration coefficient could occasionally be capturing the combined effect

of household migration and remittance receipts, which are likely to work in opposite directions. Specifically, remittances are expected to ease investments in education by decreasing liquidity constraints. Additionally, household migration is thought to disrupt family life in ways that might impede educational investments or reduce the anticipated returns to the said educational investments. Considering that these two effects are expected to have opposite impacts on children's schooling, one has to understand that the results will show the impact of both.

In the estimation below, the equations used are based on the discussion above about the impact of easing the constraint on the schooling and other similar attributes of children such as accumulated level of schooling, the number of days spent last week in school, number of hours spent on household chores, and dropouts among children. The entire estimation procedure is repeated to measure the impact of migration on all the attributes mentioned above on age groups 5–17, 5–11, and 12–17, respectively, to determine whether migration affects older and younger children differently.

4.2. Specification Issues

Simple OLS estimations would give us biased coefficients for several reasons. First, there are several other children's and household characteristics that are unobserved—such ability and income shocks— and would result in omitted-variable bias. Second, certain characteristics of households and children could be correlated with the decision to migrate, resulting in an endogeneity problem. Third, if, while comparing socioeconomic and demographic characteristics, we were to establish that migrants differed from nonmigrants in terms of income, we would need to take corrective measures. The solutions to each of these problems are as follows.

4.2.1. Child's (Unobserved) Characteristics

As mentioned above, unobserved characteristics such as a child's ability would result in omitted-variable bias. To resolve this problem, we use parents' education as proxies. Dummy variables are used to indicate, respectively, if a child's father or mother has completed primary, secondary, or tertiary education.

4.2.2. Endogeneity of Migration Variable

The second and most important concern is the endogeneity of the migration variable. Certain variables omitted from the equation—such as market conditions—simultaneously affect the decision to migrate and send children to school; these are absorbed by the error term. Different authors have used different instrumental variables to resolve this problem. Bansak and Chezum (2009) use past literacy rates and political unrest, which suggests that, historically, districts with higher literacy rates have better job prospects due to agglomeration economies and, thus, individuals in these districts are less likely to migrate. Likewise, districts with political unrest have disrupted social networks. The presence of Western Union branches within a district during past years has also proved an important instrument for analysis (Amuedo-Dorantes & Pozo, 2007).

Acosta (2006) uses current migration rates calculated as the migration propensity of the county/village of residence to eliminate endogeneity. Mansuri (2006) uses the proportion of households that currently have a family member abroad—in the area of reference—as an instrument for migration. A better instrument suggested by Hanson and Woodruff (2003) is the historic migration rate of each village. Since some regions are more accustomed to sending migrants abroad for reasons other than income diversification—such as political unrest—this activity decreases the cost of future migration by establishing informal linkages between emigrants and their area members by decreasing the information barrier.

In this working paper, we use district-level data on historic migration rates in Punjab for the period 1980–2000 to create instruments, since previous migration networks facilitate current moves. These rates vary from one district to the next, but are constant for all households living in that district. One solution to the problem of constant rates across households in one district is for the historic migration rates to interact with a particular household characteristic that facilitates migration. Hanson and Woodruff (2003) are of the view that allowing district fixed effects to interact with mothers' education could explain these variations in the household's decision-making process—the better educated a mother, the less likely her husband is to migrate.

Mansuri (2006), however, argues that having the number of adult males in the household interact with the historic migration rate—which does not vary across districts—would prove a better instrument since households with more adult males would face fewer security issues, in turn facilitating migration. She allows the historic migration rate to interact with the number of adult males in the household since the mother's education is highly correlated with the child's enrolment, which could also create an endogeneity problem.

4.2.3. Missing Families

Hanson and Woodruff (2003) raise the issue of "missing families," confining their analysis to rural areas where emigrants are more likely to leave their families behind. Urban emigrants tend to take their families with them, which results in missing families in the data (see Durand, Massey, and Zenteno, 2001; Marcelli & Cornelius, 2001; Massey, Arango, Hugo, Kouaouci, Pellegrino, & Taylor, 1993). Our data suggests that there are a significant number of urban households with external migrants. This means that missing families in urban areas do not create a problem and there are enough urban households with migrants to provide sufficient variation in the dataset to run regressions. We therefore conduct our analysis for both rural and urban areas.

A possible explanation for why Pakistan does not face the problem of missing families to as great an extent is that the majority of its labor force emigrates either to the United Arab Emirates or to developed countries such as the US or UK. The strict labor laws in the Emirates restrict individuals from taking their families with them, while many migrations to developed countries are illegal because of the high costs associated with it. These problems create barriers for migrants who might otherwise want to take their families with them.

4.2.4. Household Characteristics

Migrants' household characteristics, when compared to nonmigrants' households, show that there is a distinct difference between the two groups' income levels—migrants' households do not act as a random sample. We therefore have to add an income variable to the equation, but this variable acts as an endogenous variable since households make decisions concerning schooling and migration simultaneously, based on their income. This problem can be dealt with either by including an asset index (see Acosta, 2006) or simply by adding those assets the household already possesses, i.e., assets that do not depend on present

income and of which the migration decision is independent. We use both approaches in this working paper.

4.2.5. The Model with Specification Issues Resolved

Equation (8), therefore, uses historic migration rates—interacting with the number of adult males in the household—to instrument for migration. Similarly, in order to address the endogeneity of the income variable in the specification, we add household assets to the final equation, and derive the following econometric equation:

$$S_{ghi} = \beta_1 \widetilde{M}_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{hi} + \mu_{ghi}$$
(9)

 S_{ghi} is the schooling outcome, C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, B_{hi} is the child's gender and \widetilde{M}_{ghi} represents the fitted values of migration from the first-stage regression.

4.3. Data Sources

The data for this analysis has been taken from the Government of the Punjab's (2009) Multiple Indicator Cluster Survey for 2007/08—a household-level dataset for Punjab that is representative at the district and *tehsil* (district subdivision) level and comprises 91,075 households. We have also used unpublished data from the Bureau of Emigration and Overseas Employment in Pakistan (2009) to calculate historic migration rates by taking the average number of emigrating individuals at the district level and dividing this by the population in those districts. The historic migration rate then interacts with the number of adult males in each household in the given district to create an instrumental variable that varies at the household level, where the number of males was taken from the Multiple Indicator Cluster Survey.

4.4. Summary Statistics

4.4.1. Average Number of Children Going to School

The summary statistics reveal that there are some variations in the school level attended by children based on whether or not they come from migrants' households. Figure 2 shows that about 12 percent of children who come from households with migrants are enrolled in preschool with about the same figure for children who come from households without migrants. While about 50 percent of children from migrants' households attend primary school, the corresponding figure for children from households without migrants is 56 percent. Enrolments at matriculation level are almost 13 percent for children from households with migrants but 9 percent for children from households without migrants.

Figure 2: Percentage of children attending different levels of school in Punjab



Source: Authors' calculations based on data from Multiple Indicator Cluster Survey (Government of the Punjab, 2009) and Bureau of Emigration and Overseas Employment (2009).

4.4.2. Household Mothers' Level of Education

Figure 3 shows how, on average, mothers' education varies significantly: Nearly 43 percent of mothers from households with migrants never enrolled in school as compared to 63 percent of mothers from households with migrants. As far as primary education is concerned, 21 percent of mothers who attended primary school came from families with migrants as compared 15 percent who did not.

The rate for mothers' education in families with migrants is 11, 16, and 9 percent for those who completed middle, secondary, and higher education, respectively. The corresponding figures for families without migrants are 7, 8, and 6 percent, respectively. This shows that, on average, mothers from households with migrants are more likely to be educated than those from households without migrants. The low percentage of 42 percent for mothers—from migrants' households—who have never attended school, relative to 63 percent for those from nonmigrants' households strengthens the argument that external migrants' households have better-educated mothers.



Figure 3: Education level of household mothers in Punjab

Source: Authors' calculations based on data from Multiple Indicator Cluster Survey (Government of the Punjab, 2009) and Bureau of Emigration and Overseas Employment (2009).

4.4.3. Household Heads' Level of Education and Profession

As Figure 4 shows, on average, almost 41 percent of household heads in migrants' households are unemployed, searching for a job, or engaged in domestic chores—that is, they have no source of primary income. In households without migrants, this figure drops to almost 10 percent.

If the household head is a government employee or is associated with a proper job or business, the chances of his/her having a migrant in the household is quite low—2.8 and 3.7 percent, respectively. This, in turn, supports the argument that the reason for an individual to migrate is to enhance his/her household's resources and decrease its constraints to credit.



Figure 4: Education level of household heads in Punjab

Source: Authors' calculations based on data from Multiple Indicator Cluster Survey (Government of the Punjab, 2009) and Bureau of Emigration and Overseas Employment (2009).

4.4.4. Average Enrolment of Children by Gender

As Figure 5 shows, girls from households with migrants have a significantly lower enrolment rate than that of boys. The enrolment of boys and girls from households without migrants is, however, not significantly different from one another. These statistics reinforce the argument that girls in migrants' households are likely to share in domestic chores at the cost of poor enrolment rates.



Figure 5: Average enrolment of children by gender in Punjab

4.4.5. Average Enrolment in Public and Private Schools

The type of school is also very important in determining whether or not a child is likely to be enrolled. Figure 6 shows that 63 percent of enrolled students attend public, i.e., state-run, schools, while 37 percent go to private schools. About 42 percent of children from migrants' households in urban areas attend public schools, compared to 47 percent from those of non migrants. The corresponding figures for children attending private school are 58 and 54 percent, respectively.



Figure 6: Average enrolment in public and private schools in Punjab

Source: Authors' calculations based on data from Multiple Indicator Cluster Survey (Government of the Punjab, 2009) and Bureau of Emigration and Overseas Employment (2009).

Source: Authors' calculations based on data from Multiple Indicator Cluster Survey (Government of the Punjab, 2009) and Bureau of Emigration and Overseas Employment (2009).

The proportion of children enrolled in private school in rural areas is very low among those from households with and without migrants—46 and 26 percent, respectively, and 54 and 74 percent in public schools. The statistics show that, in both rural and urban areas, a high percentage of households with migrants send their children to private schools, whereas a comparatively high percentage of households without migrants send their children to public schools.

Another important point revealed by the statistics is that a high percentage of rural households send their children to public schools regardless of whether or not they have a migrant in their families. This could be accounted for by the low cost of schooling associated with public schools in terms of school fees and other factors.

4.4.6. Households' Wealth Index Quintile

For households, social status is also important when making the decision to migrate or to enroll a child in school. We use a wealth index to eliminate the endogeneity problem related with the household income variable—otherwise, any shocks in income can simultaneously affect the decision to migrate and enroll a child.

The wealth index is divided into five quintiles, from the lowest (poorest) to the highest (richest). Figure 7 shows that 12 percent of households without migrants fall in the lowest quintile of the wealth index. The percentage of households that fall in the lowest quintile decreases substantially to 2 percent for households with migrants. Among households without migrants, 19 percent fall in the second quintile whereas this figure is only 7 percent for households with migrants. The corresponding figures for the third quintile are 23 and 16 percent, for households without and with migrants, respectively.



Figure 7: Percentage of households in different wealth index quintiles in Punjab

Source: Authors' calculations based on data from Multiple Indicator Cluster Survey (Government of the Punjab, 2009) and Bureau of Emigration and Overseas Employment (2009).

The fourth and fifth quintiles represent the higher-income class of society. The statistics show that the collective majority of households with at least one external migrant lie in the fourth and fifth quintiles. About 36 percent of households with migrants lie in the fourth quintile; the figure rises to 41 percent in the fifth quintile. In the highest quintile, households without migrants account for 23 percent and for 23 percent in the fourth quintile. These statistics show that households with migrants and receiving remittances have access to greater resources and, hence, a better standard of living.

5. Estimating the Effect of Historic Migration Rates on Current Migration

The econometric specification we follow uses all the possible variables mentioned above that could affect the household's schooling decisions concerning its children. In the first stage, we estimate the impact of historic migration rates—interacted with the number of adult males in the household—on current migration. Note that this first stage remains the same for all the instrumental variable estimations mentioned in the previous section.¹ A priori, the current migration rate should be positively related to regional migration networks—i.e., the historic

¹ Regressions were run to check if there was a significant relationship between current school outcomes and the instrumental variable used for the analysis, because a good instrument would require that there be no relationship with the dependant variable. The results show that historic migration rates do not have any significant impact on current enrolments.

migration rate calculated at district level—since these networks strongly influence the current decision-making process of adults in those regions. This is because household characteristics are positively and consistently influenced by the gains from migration.

Table 1 gives the results for the first-stage regression. The migrant dummy is kept as a binary dependent variable, and district-level average historic migration rates for the period 1980–2000 interacted with the number of adult males in the household as the key explanatory variable in the specification. We also add to the specification all other variables that could influence the decision to migrate at the household level. These include the household head's age and age-squared, wealth indices, the household head's marital status and level of education, proportion of women working, and so forth.

The results from the simple OLS regressions are reported in the first and second columns of the table, which show that historic migration rates have a significant and positive relationship with having an external migrant in the household. The only difference between the two columns is that, in the OLS (1) specification, we add wealth indices to the specification, which deals with the fact that households with and without migrants are not random samples and, hence, variables that can distinguish between these households' social status and wealth need to be added. In the OLS (2) specification, the assets that each household possesses—rather than wealth indices—are added directly to the specification. The two specifications give approximately the same results and show a significant, positive relationship between the instrument and migration. The third column presents the results yielded by a probit analysis, where the coefficients are the marginal effects. These results reinforce the earlier result that the instrument is positively related to migration.

The other explanatory variables added to the specification also have the same signs as expected. The rural dummy has a significant, positive coefficient, indicating that individuals in rural areas migrate more so than those in urban areas because of their desire to improve their standard of living. The dummies for the wealth indices indicate that households who are among the higher-score indices tend to have an external migrant. The fourth column shows that individuals with better facilities such as a television, electricity, and so forth, tend to have an external migrant in their household. The proportion of women working in a household decreases significantly if there is an external migrant in their household. This is also fairly consistent with the literature: If the proportion of women working in a household is higher, the probability of their having an external migrant in the household falls significantly as the need for migration decreases.

The household head's level of education and profession play an important role in determining whether male members are likely to choose to migrate. The results show that, if the household head has attained primary education, then there is a significantly high probability of having an external migrant in the household than in those with an uneducated head. The literature argues that better-educated household heads imply a greater awareness among such households of the opportunities available abroad; they would be more likely to send their family members abroad. However, migration in households whose heads have attained higher education is significantly lower than in those headed by uneducated individuals. This can be explained by the argument that highly educated household heads tend to have better opportunities in their home countries and, thus, are less likely to need to improve their standard of living through migration.

As the household head's age increases, the probability of his/her sending family members abroad decreases, but this effect diminishes as the person ages further. The coefficients' direction in the probit and OLS models was generally consistent with the literature. The dummy variables added for the profession of the household head in the regression indicate that, if the household head is a government employee or owns a private business, then the possibility of his/her household having an external migrant is significantly lower than that of those households with an unemployed head.

Table 1: Estima	ting the effect of I	nistoric migration ra	tes on the migrant d	ummy
Dependent variable = 1 if there is an external migrant in the household	OLS (1)	OLS (2)	Probit (1)	Probit (2)
Number of adult males * average historic migration rate	4.761653***	6.686588***	1.821809***	2.804213***
5	(0.471946)	(0.4809517)	(0.2564573)	(0.2721793)
Rural = 1	0.0519144*** (0.0038479)	0.0279441*** (0.0040673)	0.0170851*** (0.0015831)	0.0117522*** (0.0021143)
Wealth index 2	0.0176122***		0.0621742***	
	(0.0035141)		(0.0089763)	
Wealth index 3	0.0286561***		0.0794907***	I
	(0.0035282)		(0.0090346)	
Wealth index 4	0.0848196***		0.165182^{***}	I
	(0.0038801)		(0.0134478)	
Wealth index 5	0.1546896^{***}		0.3205208***	I
	(0.0053593)		(0.0240796)	
Married household head	0.080123***	0.0826849***	0.0181035***	0.0198425***
	(0.0045909)	(0.0046583)	(0.0014853)	(0.0015013)
Value of agricultural land owned	4.51e-12	2.92e-12	2.76e-12	2.41e-12
	(5.16 e-12)	(5.19 e-12)	(2.71 e-12)	(2.78 e-12)
Number of rooms in the house	0.0013146	0.0045243***	0.0005504	0.0024897***
	(0.0010488)	(0.001065)	(0.0006638)	(0.0006995)
Value of house owned	1.40 e-09**	2.00 e-09***	6.75 e-10**	8.33 e-10***
	(5.57 e-10)	(5.64 e-10)	(2.70 e-10)	(2.78 e-10)
Household head's gender (male = 1)	-0.3998146^{***}	-0.4042105***	-0.3590238^{***}	-0.383718^{***}
	(0.0060193)	(0.0060714)	(0.0162192)	(0.0163884)

Rabia Arif and Azam Chaudhry

Continued...

Dependent variable = 1 if there is				
an external migrant in the household	OLS (1)	OLS (2)	Probit (1)	Probit (2)
Household head's age	-0.0071826***	-0.0072438^{***}	-0.0026739***	-0.0028387***
	(0.0005137)	(0.0005294)	(0.0003294)	(0.0003466)
Household head's age-squared	0.0000845^{***}	0.0000863^{***}	0.0000324^{***}	0.0000352**
	(4.86e-06)	(5.01e-06)	(3.01e-06)	(3.16e-06)
Household head's level of education	0.0108956^{***}	0.0138287***	0.0061144^{**}	0.0104277***
if primary				
	(0.0033522)	(0.0033895)	(0.0026067)	(0.0029049)
Household head's level of education if middle	0.0026338***	0.0048465	0.0013797	0.00525*
2	(0.0034486)	(0.0034802)	(0.0025427)	(0.0028565)
Household head's level of education	-0.0054356^{***}	-0.0014986	-0.0033749	5.38e-06
if secondary				
	(0.0031403)	(0.0031655)	(0.0021278)	(0.0023574)
Household head's level of education if higher	-0.0257753***	-0.0169949***	-0.0103228***	-0.0080037***
D	(0.0042815)	(0.0043674)	(0.0023596)	(0.0027347)
Household head if government	-0.0244116***	-0.0258793***	-0.0141716***	-0.0141164***
employee				
	(0.0040563)	(-0.0041289)	(0.0021008)	(0.0022669)
Household head if private employee	-0.0036568	-0.004247	0.0018	0.0021113
	(0.0045842)	(0.0046507)	(0.0032817)	(0.0035149)
Household head if agriculturist	-0.0074228^{***}	-0.0076244^{***}	-0.0056748***	-0.0057853^{***}
	(0.002553)	(0.0025947)	(0.0017622)	(0.0018601)
Household head if livestock owner	0.0019949	0.0049183	0.0097276*	0.0117848*
	(0.0081649)	(0.0082755)	(0.0063511)	(0.0069138)
				Continued

Continues...

:	
S.	
ĭ	
.E	
Ē	
Ö	

Dependent variable $= 1$ if there is				
an external migrant in the household	OLS (1)	OLS (2)	Probit (1)	Probit (2)
Total number of household members	-0.0078246^{***}	-0.0093831***	-0.003206***	-0.0042923^{***}
	(0.0006618)	(-0.0006805)	(0.0004518)	(0.0004898)
Total number of women (aged 15-49)	0.040853^{***}	0.0429469***	0.0192612***	0.0215425^{***}
	(0.0026921)	(0.0027306)	(0.0017044)	(0.0017809)
Proportion of working women	-0.0147761^{***}	-0.0137542^{***}	-0.0077925***	-0.0071589^{***}
	(0.0026179)	(0.0026655)	(0.0021691)	(0.002267)
Total number of infants	0.0091721***	0.0090525***	0.004376^{***}	0.004421^{***}
	(0.0012492)	(0.0012756)	(0.000832)	(0.000883)
Total number of children (aged 5–17)	0.0016508^{*}	0.0024494***	-0.0012317**	-0.0007727
	(0.0008836)	(0.0009022)	(0.0006107)	(0.0006489)
Constant	0.4107908***	0.4381186***	I	I
	(0.0145192)	(0.0152258)		
Number of observations	41,289	40,431	41,289	40,431
R-squared	0.1781	0.1717	I	I
Adjusted R-squared	0.1776	0.1711	I	I
Pseudo R-squared			0.259	0.2511
F-statistic (to test joint significance of	343.86	253.82	ı	I
excluded instruments)				

Notes: * = p < 0.10, $*^* = p < 0.05$, $*^* = p < 0.01$. Standard errors are given in parentheses below estimates. In the OLS (2) and probit (2) columns, the assets that households use were added directly to the regression equation to incorporate the wealth effect. The coefficients indicate that both estimations yield the same results. Source: Authors' calculations.

6. Estimating the Impact of External Migration on School Enrolment, Accumulated Level of Schooling, and Days Spent in School Last Week

6.1. Estimating the Impact of External Migration on School Enrolment

In the first part of our analysis, we estimate the impact of external migration on a child's school enrolment, using the equation

$$S_{ghi} = \beta_1 \widetilde{M}_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{hi} + \mu_{ghi}$$
(10)

where S_{ghi} is the schooling outcome, C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, B_{hi} is

the child's gender, and \widehat{M}_{ghi} represents the fitted values of migration from the first-stage regression.

We first focus on children aged 5–17. Then, to test whether migration affects younger children differently, we divide the sample into two subsamples. Table 2 gives the results for the first hypothesis. The simple OLS results are given in columns 2, 6, and 10. The "IV reg" columns (3, 7, and 11) indicate those results yielded using an instrumental variable, while the "IV probit" column (4) gives the results yielded when a probit model is used with the instrumental variable to estimate the coefficients. District dummies are added to the IV probit with fixed effects in columns 8 and 12.

The results for the 5–17 age group show that external migration has a positive, significant relationship with school enrolments for children aged 5–17. This means that more children from households with migrants are enrolled in school than those from households without migrants. Columns 5, 9, and 13 give the results of the same analysis but without incorporating district fixed effects; in this regression equation, the external migrant coefficient becomes negative and insignificant.

The urban dummy becomes negative and significant if district fixed effects are eliminated from the specification, indicating that rural enrolment is high; this could be because of the higher rural population compared to urban areas. The gender dummy—which takes the value of 1 if the child is male—has a positive, significant coefficient, which means that boys' enrolment is higher than that of girls, implying a gender bias. Likewise, the number of siblings of all ages is negatively related to school enrolments, which is also consistent with the literature.

The results for younger children—aged 5–11—indicate that external migration has a positive, significant impact on their school enrolment. All the other variables added to the specification show the same signs as they did for the first hypothesis and are consistent with the literature. For older children, we obtain the opposite results: The coefficient associated with the instrumental variable is not significant. Hence, external migration does not significantly affect the enrolment of older children. A plausible explanation for this result is that current migrations make it difficult for older children to enroll in that time period.

Dependent		Age grot	up 5–17			Age gr	oup 5–11			Age gro	up 12–17	
variable: Child onrolled				IV probit			/ neohit with	IV probit			IV neobit	IV probit
currently	OLS	IV reg	IV probit	district FE	OLS	IV reg	district FE	district FE	OLS	IV reg w	ith district FE	district FE
-	2	3	4	ъ	9	7	8	6	10	11	12	13
External migrant in household	0.187***	-0.165	4.079***	-1.107	0.0783	-0.502	5.028***	-0.955	0.203***	-0.0198	1.481	-1.590
Household hood's	(0.0503)	(0.472)	(1.411) 0.0220***	(0.932) 0.00600	(0.130)	(0.686) 0.00100	(0.896) 0.0189***	(2.035)	(0.0540)	(0.346) 0.00118	(1.316)	(1.475)
rrouserioiu rieau s age	71100	+01000	0770.0	000000	0110.0-	06100.0-	60107-	0010.0-	7/1000.0-	01100.0-	/ /70000	I
Household head's	(0.00379) -0.0000964***	(0.00176) -0.00000729	(0.00380) -0.000235***	(0.00530) -0.0000351	(0.0103) 0.000123	(0.00216) 0.0000234	(0.00574) 0.000229***	(0.0114) 0.000171	(0.00416) -0.000137***	(0.00184) -0.0000334*	(0.00628) -0.000193***	(0.00889) -0.0000399
age-squared												
Household head's	(0.0000366) 	(0.0000205)	(0.0000470) 1.082**	0.0000566)	(0.000101)	0.0000273)	(0.0000554) _1 270***	(0.000122)	(0.0000402) -0.134**	0.0000191)	(0.0000664) 0.210	(0.0000917)
gender = 1 if male			700.1		001.00	10.00	6/71-				6170	
5	(0.0510)	(0.137)	(0.448)	(0.280)	(0.125)	(0.214)	(0.288)	(0.675)	(0.0550)	(0.0952)	(0.372)	(0.408)
Proportion of	-0.0667***	-0.0147**	0.00738	-0.0905***	-0.103***	-0.0121	-0.115***	-0.141***	-0.0585***	-0.0186***	-0.0398	-0.0839***
working women	Ĩ						00000	10000				
Child's age	(cc10.0) -0.122***	(0.00696) 0.0788***	(0.0394) -0.0902***	(0.0181) -0.116***	(0.0357) 0.598***	(0.0103) 0.0283***	(0.0239) 0.274	(0.0397) 0.581***	(0.01/2) -0.443***	(0.0606**	(0.02/0) -0.411***	(0.0221) -0.438***
)	(0.0235)	(0.00207)	(0.0346)	(0.0236)	(0.101)	(0.00411)	(0.171)	(0.108)	(0.105)	(0.0263)	(0.113)	(0.103)
Child's age-squared	-0.00625***	-0.00514***	-0.00437** (0.00177)	-0.00610***	-0.0463***	-0.00209*** 0.000361)	-0.0214*	-0.0449*** 0.00678)	0.00519	-0.00473*** 0.000916)	0.00434 (0.00369)	0.00561
Child's gender = 1 if male	0.0992***	0.00946***	0.0652**	0.0973***	0.0683	0.00278	0.0463	0.0662	0.115***	0.0283***	0.109***	0.111***
	(0.0186)	(0.00245)	(0.0310)	(0.0184)	(0.0424)	(0.00225)	(0.0302)	(0.0413)	(0.0212)	(0.00537)	(0.0226)	(0.0219)
Total infants in household	-0.0430***	-0.00363	-0.0604***	-0.0300**	0.00333	0.00438	0.0461***	0.00935	-0.0560***	-0.0138***	-0.0631***	-0.0383**
Total bovs aged 5–	(0.00878) -0.0198*	(0.00372) -0.00528**	(0.00688) -0.0293***	(0.0126) -0.0143	(0.0195) -0.0654***	(0.00584) -0.000144	(0.0122) -0.00453	(0.0265) -0.0586**	(0.00998) -0.00880	(0.00347) -0.00182	(0.0114) -0.0128	(0.0181) -0.00159
9 in household	(0.0107)	(0.00.23)	(0.00843)	(0.0112)	0 0223)	(0.00335)	(0.0.26.2)	(0.0758)	(0.01.25)	0 00335)	(0.01.29)	(0.0134)
					, ,						Ŭ	ontinued

Table 2: Estimating the effect of external migration on school enrolments

Dependent		Age grou	up 5–17			Age gi	roup 5–11			Age gro	oup 12–17	
variable:		0		IV probit		5	IV probit	IV probit		0	-	IV probit
Child enrolled				without			with district	without			IV probit	without
currently	OLS	IV reg	IV probit	district FE	OLS	IV reg	Æ	district FE	OLS	IV reg	with district FE	district FE
1	2	3	4	5	9	7	8	6	10	11	12	13
Total girls aged 5–9 in household	-0.0313***	-0.00601***	-0.0290***	-0.0282**	-0.0467**	0.000199	-0.00493	-0.0419*	-0.0362***	-0.00868**	* -0.0333**	-0.0353***
	(0.0111)	(0.00160)	(0.0103)	(0.0110)	(0.0238)	(0.00284)	(0.0201)	(0.0250)	(0.0128)	(0.00330)	(0.0133)	(0.0125)
Total boys aged 10–17 in hourohold	-0.0398***	-0.00451***	-0.0248*	-0.0405***	-0.0354*	-0.000491	-0.0121	-0.0333*	-0.0367***	-0.00999**	* -0.0308***	-0.0438***
	(0.00804)	(0.00114)	(0.0131)	(0.00783)	(0.0183)	(206000:0)	(0.0145)	(0.0182)	(0.00911)	(0.00268)	(0.0115)	(0.00935)
Total girls aged 10– 17 in household	-0.00264	-0.000603	0.00622	-0.00519	0.0193	0.000217	0.00298	0.0198	0.000848	-0.00121	0.00544	-0.00837
	(0.00814)	(0.00143)	(0.00737)	(0.00824)	(0.0190)	(0.00107)	(0.0122)	(0.0189)	(0.00923)	(0.00273)	(0.0106)	(0.0112)
Household head if govt. emplovee	0.0926***	0.00798	0.155***	0.0635	-0.0728	-0.0119	-0.148***	-0.0941	0.122***	0.0222*	0.151***	0.0742
	(0.0327)	(0.0114)	(0.0270)	(0.0402)	(0.0807)	(0.0145)	(0.0436)	(0.0861)	(0.0354)	(0.0118)	(0.0433)	(0.0585)
Household head if	0.0324	0.00284	0.0479**	0.00995	0.0554	-0.00173	0.0136	0.0325	0.0301	0.00600	0.0367	0.00457
private employee	0.0284)	(0 00463)	(0,0,0,0)	(00000)	0.0734)	(00000)	00456)	(0.0745)	(0.0310)	(0.0078.4)	(0.0311)	(0.0324)
Household head if	-0.0731***	-0.0117**	-0.0238	-0.0863***	0.0701	-0.00620	0.0807**	-0.0842	-0.0815***	-0.0266***	-0.0721**	-0.0921***
laborer												
	(0.0233)	(0.00463)	(0.0345)	(0.0231)	(0.0555)	(0.00602)	(0.0320)	(0.0554)	(0.0260)	(0.00705)	(0.0284)	(0.0252)
Household head's education if primary	0.000930	0.00000169	-0.0231	0.0127	0.110**	0.00699	0.0781*	0.112**	-0.0187	-0.00536	-0.0258	-0.00124
	(0.0226)	(0.00422)	(0.0201)	(0.0231)	(0.0559)	(0.00465)	(0.0414)	(0.0549)	(0.0252)	(0.00701)	(0.0258)	(0.0277)
Household head's education if middle	0.0981***	0.0151***	0.0551	0.111***	0.0494	0.00458	0.0524	0.0602	0.111***	0.0306***	0.106***	0.122***
	(0.0259)	(00390)	(0.0368)	(0.0253)	(0.0628)	(0.00430)	(0.0364)	(0.0621)	(0.0286)	(0.00738)	(0.0294)	(0.0280)
Household head's education if	0.253***	0.0330***	0.183***	0.261***	0.0285	0.00294	0.0293	0.0391	0.297***	0.0736***	0.295***	0.290***
secondary	070740	(0.00318)	0.0667	(0.0250)	(0.0588)	(0.00.78)	0.03.43)	0.0573)	(0.0273)		0.0788)	00415)
	(0 - 200)	(01,000,0)	(100000)	(00000)	(000000)	(0.0000)	(01000)	11 1000	(0.1200)	(000000)	(00700)	

variable: IV probit Child enrolled IV probit without Currently OLS IV reg IV probit district FE OLS Currently OLS IV reg IV probit district FE OLS Household head's 0.481*** 0.0418*** 0.358*** 0.482*** 0.258 Household head's 0.481*** 0.00443 (0.121) (0.0422) (0.093 Fenale bias 0.03599 0.01079 0.04430 (0.121) (0.0422) (0.0057) Wealth index 0.02533 (0.01077) (0.0550) (0.0542) (0.0054) Wealth index 0.251*** 0.03022 (0.116*** 0.2571) (0.054) Wealth index 0.251*** 0.03022 (0.024) (0.0251) (0.0251) Wealth index 0.251*** 0.03022 (0.02124) (0.05151) (0.0551) Wealth index 0.251*** 0.03022 (0.0251) (0.0251) (0.05151) (0.0551) <th>IV probit withour bit district FE 5 *** 0.482*** 0</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>	IV probit withour bit district FE 5 *** 0.482*** 0								
	without bit district FE 5			IV probit	IV probit				IV probit
currently OLS IV reg N probit district FE OLS 1 2 3 4 5 6 Household head's 0.481*** 0.0418*** 0.358*** 0.482*** 0.258 Household head's 0.481*** 0.0118*** 0.358*** 0.482*** 0.258 education if higher 0.0369 0.00443 0.121 0.0422 0.093 Female bias 0.03239 0.0119 0.0440 0.0665 0.033 Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.209 Wealth index 3 0.251*** 0.0330*** 0.00570 0.0271 0.054 Wealth index 3 0.251*** 0.0330*** 0.03129 0.01124 0.03119 0.054 Wealth index 3 0.251*** 0.0330*** 0.00315 0.055 Model hidex 3 0.251*** 0.03319 0.05315 0.055	bit district FE 5		7	<i>vith</i> district	without			IV probit	without
1 2 3 4 5 6 Household head's 0.481*** 0.0418*** 0.358*** 0.482*** 0.258' education if higher (0.0369) (0.00443) (0.121) (0.0422) (0.093 Female bias 0.03059) (0.0107) (0.0550) (0.0053) (0.003 Wealth index 2 0.108*** 0.0125* 0.03020 0.116*** 0.209 Wealth index 3 0.251*** 0.00672) (0.00570) (0.0243) (0.10540) (0.0542) (0.00540) Wealth index 3 0.251*** 0.03020 0.116*** 0.209 (0.0540) (0.0560) (0.0560) <t< th=""><th>5 *** 0.482*** 0 (0.0422) (0</th><th>OLS</th><th>IV reg</th><th>Æ</th><th>district FE</th><th>OLS</th><th>IV reg 1</th><th>with district FE</th><th>district FE</th></t<>	5 *** 0.482*** 0 (0.0422) (0	OLS	IV reg	Æ	district FE	OLS	IV reg 1	with district FE	district FE
Household head's 0.481*** 0.0418*** 0.358*** 0.482*** 0.258 education if higher (0.0369) (0.0043) (0.121) (0.0422) (0.093 Female bias -0.03593 -0.0119 0.04410 0.0665 -0.039 Female bias -0.03523) (0.0107) (0.0550) (0.0542) (0.101) Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.201 Wealth index 3 0.251** 0.00672) (0.0242) (0.0167) (0.0542) (0.101) Wealth index 3 0.251*** 0.0302 0.116*** 0.2711 (0.0542) (0.0542) (0.0542) (0.0542) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542) (0.0101) (0.0542)	*** 0.482*** 0	9	7	8	6	10	11	12	13
Termale bias (0.0369) (0.00443) (0.121) (0.0422) (0.093) Female bias -0.0398 -0.0119 0.0440 -0.0605 -0.039 Wealth index 2 0.108*** 0.0107) (0.0550) (0.0542) (0.101 Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.209 Wealth index 3 0.2059 (0.00672) (0.0498) (0.0271) (0.0240) Wealth index 3 0.251*** 0.3032 0.116*** 0.209 Wealth index 3 0.251*** 0.00672) (0.0498) (0.0254) (0.0254) Wealth index 3 0.251*** 0.00672) (0.0498) (0.0271) (0.054) Wealth index 3 0.251*** 0.0326** 0.03155 (0.0557) (0.0577)) (0.0422) (0	258*** 0	0.00225	0.114	0.259***	0.516***	0.101***	0.504***	0.506***
Female bias -0.0398 -0.0119 0.0440 -0.0665 -0.039 Wealth index 2 0.108*** 0.0107) (0.0550) (0.0542) (0.101 Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.209 Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.209 Wealth index 3 0.251*** 0.0302* (0.0271) (0.054 (0.054 Wealth index 3 0.251*** 0.0302** 0.0302 0.116*** 0.209 Wealth index 3 0.251*** 0.0302** 0.0315 (0.054 Vealth index 3 0.251*** 0.0315 (0.054 0.02288 0.0124 (0.10315) (0.0515))) (20032)	0.00483)	(0.0942)	(0.0946)	(0.0398)	(0.00872)	(0.0474)	(0.0614)
(0.0523) (0.0107) (0.0550) (0.0542) (0.101 Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.209 Wealth index 2 0.0269) (0.00672) (0.0271) (0.0574) (0.0574) Wealth index 3 0.251*** 0.0305** 0.0498) (0.0271) (0.054) Wealth index 3 0.251*** 0.0310*** 0.0315) (0.05315) (0.06315)	0.0605 -0)0393	0.0133	-0.0869	-0.0724	-0.0120	-0.00355	0.0112	-0.0387
Wealth index 2 0.108*** 0.0125* 0.0302 0.116*** 0.209 Wealth index 3 0.0269) (0.00672) (0.0498) (0.0271) (0.0574) Wealth index 3 0.251*** 0.0330*** 0.0802 0.271*** 0.357 Wealth index 3 0.251*** 0.0330*** 0.0802 0.271*** 0.055 Wealth index 3 0.251*** 0.0124) (0.0315) (0.06315) (0.06210)	0) (0.0542) (0)) (101) (I	0.0170)	(0.0584)	(0.111)	(0.0601)	(0.0164)	(0.0640)	(0.0648)
(0.0269) (0.00672) (0.0498) (0.0271) (0.054 Wealth index 3 0.251*** 0.0330*** 0.0802 0.271*** 0.357 Wealth index 3 0.251*** 0.0330*** 0.00315) (0.06315) 0.0602 Wealth index 4 0.0124) 0.10124) 0.01231) (0.06315) 0.0652	2 0.116*** 0).209*** 0	.0141**	0.159***	0.205***	0.0896***	0.0283***	0.0685*	0.109***
Wealth index 3 0.251*** 0.0330*** 0.0802 0.271*** 0.357 (0.0288) (0.124) (0.123) (0.0315) (0.063 (0.062) (0.062) (0.063) (0.063) (0.063)	8) (0.0271) (0)) (0240)	0.00714)	(0.0567)	(0.0524)	(0.0318)	(0.0100)	(0.0398)	(0.0340)
(0.0288) (0.0124) (0.103) (0.0315) (0.062	2 0.271*** 0).357*** 0	0.0252	0.286***	0.359***	0.241 ***	0.0715***	0.198***	0.278***
) (0.0315) (0	0.0620) ((0.0156)	(0.0824)	(0.0644)	(0.0333)	(0.0132)	(0.0608)	(0.0391)
	9 0.495*** 0	.508*** 0	0.0467	0.523***	0.531***	0.454***	0.130***	0.366***	0.518***
(0.0326) (0.0276) (0.202) (0.0497) (0.073) (0.0497) (0)) (92.0736)	0.0393)	(0.0963)	(0.112)	(0.0371)	(0.0224)	(0.110)	(0.0616)
Wealth index 5 0.898*** 0.121*** 0.299 0.925*** 0.874	0.925*** 0).874*** 0	0.0676	0.825***	0.866***	0.894^{***}	0.223***	0.758***	0.951***
(0.0406) (0.0409) (0.357) (0.0585) (0.101) (0.0585) (0)) (101)	0.0580)	(0.169)	(0.143)	(0.0451)	(0.0324)	(0.173)	(0.0668)
Constant 3.474*** 0.734*** 1.068 3.886*** 0.642	3.886*** 0	.642 1	.081***	1.714***	1.163	5.601 ***	0.773***	4.826***	6.147***
(0.185) (0.169) (1.416) (0.272) (0.503)) (0.272) (0	0.503) ((0.260)	(0.301)	(0.880)	(0.764)	(0.250)	(1.220)	(0.771)
Observations 69,273 69,286 58,355 69,286 39,19.	5 69,286 3	9,193 3	9,429	37,858	39,429				
R-squared . 0.194		•					0.147		
Adi, R-squared . 0.194 .		•					0.145		
Pseudo-R-squared 0.312 0.115	0	.115				0.154			

30 The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

6.2. Estimating the Impact of External Migration on Accumulated Level of Schooling

Since some economists believe that school enrolment is not an adequate measure of human capital, we also use the accumulated level of schooling acquired by a child, according to the following specification:

Accumulated level of schooling $_{ghi} = \beta_1 \widetilde{M}_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{hi} + \mu_{ghi}$ (11)

where S_{ghi} is the schooling outcome, C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, B_{hi} is

the child's gender, and \widehat{M}_{ghi} represents the fitted values of migration from the first-stage regression.

Table 3 presents the result of the regressions, with the accumulated level of schooling as the dependent variable for children from age groups 5–17, 5–11, and 12–17. The results for age group 5–17 are similar to those discussed in the previous section. Accumulated schooling appears to be affected significantly and positively by external migration. Likewise, the accumulated level of schooling in households headed by persons with a higher level of education is significantly higher than that of less educated or uneducated household heads. The same holds true for children whose households are headed by government employees compared to laborers or unemployed persons.

The results for the first subsample (aged 5–11) show that external migration does not have a significant impact on younger children when district fixed effects are added to the specification. Once these fixed effects are eliminated, however, the results yield a positive, significant coefficient for the external migrant variable. The coefficient of child age and child-age-squared is positive and significant, indicating that, as the child grows older, his/her level of accumulated schooling also increases. The distance-to-school variable only becomes significant when district fixed effects are eliminated from the specification.

The results for the older subsample (aged 12–17) are different from that of younger children. The regressions carried out without adding district fixed effects show that external migration has a significant, negative

relationship with older children's accumulated schooling. Once district fixed effects are added to the regression, the coefficient becomes positive and significant, indicating that external migration significantly improves older children's accumulated level of schooling.

Our results also show that among younger children, fewer males attain a higher level of education than females. Additionally, older children in households with a greater proportion of working women attain a significantly higher level of accumulated schooling, possibly indicating that such households are more aware of the job market and the opportunities associated with better education.

		Age group 5–17			Age group 5–1	1	A	ge group 12–17	
Dependent variable Accumulated level of schooling		IV reg with district FF	IV reg without district FF	SIC	IV reg <i>with</i> district FF	IV reg without district FF	SIO	IV reg with district FF	IV reg without district FF
External migrant in	0.0222*	3.818**	-1.137***	-0.000737	-0.0639	0.459**	0.0556***	3.225***	-4.841***
Household head's	(0.0114) 0.00796***	(1.529) 0.0214^{***}	(0.280) 0.00345**	(0.0114) 0.00429***	(1.781) 0.00409	(0.208) 0.00580***	(0.0208) 0.0120***	(0.787) 0.0253***	(1.216) -0.0103*
age Household head's	(0.00104) -0.0000606***	(0.00567) -0.000220***	(0.00150) -0.00000761	(0.00105) -0.0000300** *	(0.00570) -0.0000274	(0.00126) -0.0000490***	(0.00186) -0.0000916***	(0.00413) -0.000235***	(0.00620) 0.000149**
age-squared Household head's	(0.0000097) -0.0762***	(0.0000662) 1.029**	(0.0000162) -0.429***	(0.0000101) -0.0254*	(0.0000717) -0.0452	(0.0000133) 0.117*	(0.0000179) -0.112***	(0.0000428) 0.750***	(0.0000655) -1.503***
Proportion of	(0.0138) -0.00852*	(0.446) 0.0445**	(0.0862) -0.0314***	(0.0154) -0.00279	(0.561) -0.00373	(0.0692) -0.00589	(0.0227) -0.0113	(0.216) 0.0308**	(0.349) -0.0827***
Child's age	(0.00448) 0.115***	(0.0226) 0.112***	(0.00620) 0.116***	(0.00473) 0.334***	(0.0271) 0.334***	(0.00568) 0.334***	(0.00767) -0.00723	(0.0146) 0.0440	(0.0213) -0.0821
Child's age-squared	(0.00400) 0.00324*** (0.000178)	(0.00661) 0.00332*** (0.000290)	(0.000192) 0.00321*** (0.000192)	(0.0103) -0.00990*** (0.000637)	(0.000705) (0.000705)	(0.010.0) -0.00989*** (0.000653)	(0.0438) 0.00805*** (0.00152)	(0.00525*** 0.00625*** (0.00208)	(0.07/3) 0.0107*** (0.00269)
Child's gender = 1 male	if -0.0130*** (0.00469)	-0.0174** (0.00780)	-0.0113** (0.00506)	-0.0126*** (0.00486)	-0.0125** (0.00598)	-0.0128** (0.00499)	0.00389 (0.00905)	-0.000567 (0.0122)	0.0109 (0.0156)
Child is disabled Total infants in	0.00641 (0.0375) -0.0275***	0.0336 (0.0635) -0.0565***	-0.0120 (0.0404) -0.0189***	-0.0124 (0.0280) -0.0186***	-0.0132 (0.0355) -0.0181	-0.01.56 (0.0287) -0.0253***	0 (.) -0.0404***	0 (.) -0.0621***	0 (.) -0.00413
household	(0.00219)	(0.0122)	(0.00332)	(0.00206)	(0.0154)	(0.00288)	(0.00438)	(0.00796)	(0.0116)
									Continued

Rabia Arif and Azam Chaudhry

									Continues
		Age group 5–17			Age group 5-	1		Age group 12-1	7
Dependent variable:			IV reg			IV reg			IV reg
Accumulated level of schooling	OLS	IV reg with district FE	<i>without</i> district FE	OLS	IV reg <i>with</i> district FE	<i>without</i> district FE	OLS	IV reg with district FE	<i>without</i> district FE
Total boys aged 5–9 in household	0.0133***	-0.00111	0.0177***	0.0166***	0.0169*	0.0139***	0.00137	-0.00934	0.0192*
	(0.00271)	(0.00726)	(0.00312)	(0.00267)	(0.00873)	(0.00291)	(0.00535)	(0.00763)	(0.0101)
Total girls aged 5–9 in household	0.00904***	0.00309	0.0109***	0.0179***	0.0181**	0.0165***	-0.00361	0.000964	-0.00877
	(0.00278)	(0.00509)	(0.00300)	(0.00274)	(0.00719)	(0.00286)	(0.00554)	(0.00749)	(0.00954)
Total boys aged 10- 17 in household	-0.0238***	-0.0206***	-0.0251***	-0.0140***	-0.0140***	-0.0157***	-0.0256***	-0.0139**	-0.0453***
	(0.00206)	(0.00357)	(0.00223)	(0.00207)	(0.00250)	(0.00211)	(0.00392)	(0.00599)	(0.00848)
Total girls aged 10– 17 in household	-0.0168***	-0.00897**	-0.0192***	-0.0120***	-0.0121 ***	-0.0121***	-0.0162***	-0.00300	-0.0374***
	(0.00203)	(0.00455)	(0.00227)	(0.00202)	(0.00281)	(0.00208)	(0.00391)	(0.00617)	(0.00868)
Household head if	0.0388***	0.126***	0.0132	0.0249***	0.0236	0.0348***	0.0419***	0.122***	-0.0806**
govt. employee									
	(0.00768)	(0.0371)	(0.0104)	(0.00780)	(0.0379)	(0.00904)	(0.0137)	(0.0270)	(0.0385)
Household head if	-0.00497	0.0180	-0.0176**	0.00299	0.00262	0.00812	-0.0164	0.00189	-0.0591**
private employee									
	(0.00700)	(0.0146)	(0.00787)	(0.00696)	(0.0125)	(0.00736)	(0.0128)	(0.0177)	(0.0235)
Household head if laborer	-0.0361***	-0.00916	-0.0447***	-0.00868	-0.00917	-0.00273	-0.0865***	-0.0678***	-0.118***
	(0.00622)	(0.0148)	(0.00701)	(0.00615)	(0.0153)	(0.00654)	(0.0115)	(0.0161)	(0.0208)
Household head's education if primary	0.0309***	0.00833	0.0392***	0.0336***	0.0339***	0.0288***	0.0341***	0.0165	0.0714***
	(0.00618)	(0.0135)	(0.00698)	(0.00611)	(0.0125)	(0.00646)	(0.0114)	(0.0158)	(0.0215)
Household head's	0.0309***	0.00833	0.0392***	0.0336***	0.0339***	0.0288***	0.0341***	0.0165	0.0714***
euucanon II piinaly	(0.00618)	(0.0135)	(0 00698)	(0.00611)	(0.01.25)	(0 00646)	(0 0114)	(0.0158)	(00215)
Household hood's	(0-0000) ***00000		(00000) ***	***90000	(07-00) ***0000		(00165	0.0714**
education if primary	60000	CC000'0	76 00.0	00000	60000	0070.0		C010.0	+ 0.0
	(0.00618)	(0.0135)	(0.00698)	(0.00611)	(0.0125)	(0.00646)	(0.0114)	(0.0158)	(0.0215)
									Continued

34 The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

		Age group 5-1	7		Age group 5-	-11		Age group 12-	17
Dependent variable:			IV reg			IV reg			IV reg
Accumulated level of schooling	OLS	IV reg with district FE	without district FE	OLS	IV reg with district FE	<i>without</i> district FE	OLS	IV reg <i>with</i> district FE	without district FE
Household head's	0.0915***	0.0773***	0.0993***	0.0576***	0.0579***	0.0551 ***	0.137***	0.132***	0.161***
equeation il mique	(0.00685)	(0.0125)	(0.00754)	(0.00682)	(0.0112)	(0.00713)	(0.0125)	(0.0168)	(0.0219)
Household head's	0.154***	0.156***	0.159***	0.0612***	0.0614***	0.0591 ** *	0.273***	0.288***	0.268***
secondary									
	(0.00624)	(0.0101)	(0.00667)	(0.00622)	(0.00763)	(0.00639)	(0.0114)	(0.0157)	(0.0195)
Household head's	0.203***	0.216***	0.207***	0.0902***	0.0898***	0.0929***	0.376***	0.379***	0.395***
	(0,00796)	(0.0140)	(0.00851)	(0.00784)	(0.0127)	(0.00802)	(0.0148)	(0.0198)	(0.0250)
Female bias	-0.0313**	0.0369	-0.0557***	-0.0138	-0.0154	-0.00212	-0.0275	0.0278	-0.124**
	(0.0126)	(0.0342)	(0.0148)	(0.0121)	(0.0450)	(0.0136)	(0.0257)	(0.0369)	(0.0497)
Wealth index 2	0.0477***	0.00286	0.0579***	0.0493***	0.0499***	0.0457***	0.0856***	0.0392*	0.150***
	(0.00742)	(0.0217)	(0.00862)	(0.00699)	(0.0186)	(0.00738)	(0.0148)	(0.0228)	(0.0313)
Wealth index 3	0.125***	0.0307	0.149^{***}	0.101***	0.103^{**}	0.0921***	0.213^{***}	0.123***	0.345^{***}
	(0.00777)	(0.0400)	(0.0111)	(0.00744)	(0.0404)	(0.00899)	(0.0152)	(0.0301)	(0.0450)
Wealth index 4	0.197***	-0.0216	0.255^{***}	0.121^{***}	0.125	0.0976***	0.352^{***}	0.167***	0.624^{***}
	(0.00856)	(0.0890)	(0.0191)	(0.00827)	(0.102)	(0.0148)	(0.0165)	(0.0510)	(0.0798)
Wealth index 5	0.312***	-0.0129	0.389***	0.154^{***}	0.160	0.119^{***}	0.548^{***}	0.272***	0.929^{***}
	(0.0102)	(0.132)	(0.0259)	(0.00995)	(0.151)	(0.0195)	(0.0192)	(0.0731)	(0.111)
Constant	-0.876***	-2.233***	-0.417***	-1.661***	-1.637**	-1.802***	-0.345	-1.885***	2.114^{***}
	(0.0380)	(0.550)	(0.112)	(0.0510)	(0.679)	(0.0963)	(0.316)	(0.569)	(0.817)
Observations	68,688	68,688	68,688	39,095	39,095	39,095	29,593	29,593	29,593
R-squared	0.628	0.026	0.569	0.414	0.414	0.384	0.387		
Adjusted R-squared	0.627	0.025	0.569	0.413	0.413	0.384	0.386		

~ the variables for the distance to public and private schools were also added to the model. The coefficients' signs are consistent with the literature. **Source:** Authors' calculations. 6.3. Estimating the Impact of External Migration on the Number of Days Spent in School Last Week

In our third analysis, we look at the number of days the child spent in school in the preceding week. The equation estimated is

Days spent in school last week
$$_{ghi} = \beta_1 \widetilde{M}_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{hi} + \mu_{ghi}$$
(12)

where S_{ghi} is the schooling outcome, C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, B_{hi} is the child's gender, and \widehat{M}_{ghi} represents the fitted values of migration from the first-stage regression.

Table 4 presents the results for the regression when the number of days the child spent in school during the previous week is the dependent variable. The results indicate that, among the 5–17 age group, there is no significant difference in the number of days spent in school the preceding week between children from households with and those without external migrants. The results remain the same even if district fixed effects are eliminated from the specification. If we eliminate district fixed effects, the urban dummy becomes significantly negative, indicating that children in urban areas spend fewer days in school than those in rural areas. These results, however, should be considered in light of the fact that we only have data on one week's attendance.

The results for the subsample of younger children (aged 5–11) imply that external migration has no significant effect on the number of days they spent in school in the preceding week. The coefficient of the external migrant variable becomes significantly negative if district fixed effects are eliminated, showing that the attendance of younger children falls significantly if there is an external migrant in the household.

		Age group 5–1	7		Age group 5-1	1		Age group 12-	17
Dependent variable: No. of days spent in		IV reg with	IV reg without		IV reg with	IV reg without		IV reg with	IV reg without
External migrant in	-0.109* **	aistrict FE 5.616	district FE	-0.0457	district FE 1.204	district FE -1.135**	-0.0413	district FE 1.050	0.687
nousenoid	(0.0212)	(3.635)	(0.491)	(0.0313)	(1.085)	(0.563)	(0.0293)	(1.081)	(1.014)
Household head's age	-0.00573*** (0.00199)	0.0154 (0.0137)	-0.00473* (0.00273)	-0.00822***	-0.00179 (0.00639)	-0.00473 (0.00322)	-0.00825*** (0.00282)	-0.00275 (0.00617)	-0.000885 (0.00618)
Household head's age-	0.0000625***	-0.000186	0.0000539*	0.0000737**	0.00000677	0.0000661*	0.0000776***	0.0000202	-0.00000312
squared	(0.0000191)	(0.000160)	(0.0000292)	(0.0000287)	(0.0000653)	(0.0000342)	(0.0000269)	(0.0000632)	(0.0000635)
Household head's gender = 1 if male	0.0159	1.772	-0.125	0.0602*	0.434	-0.341*	0.0506	0.382	0.308
	(0.0262)	(1.115)	(0.159)	(0.0349)	(0.327)	(0.185)	(0.0329)	(0.330)	(0.320)
Proportion of working women	0.0156*	0.107*	0.00440	0.0128	0.0338	0.000522	0.0145	0.0327	0.0194
	(0.00869)	(0.0594)	(0.0118)	(0.0125)	(0.0224)	(0.0148)	(0.0116)	(0.0216)	(0.0219)
Child's age	0.0861*** (0.00766)	0.0846***	0.0886*** (0.00789)	0.129* (0.0699)	0.149** (0.0745)	0.0748***	0.134*** (0.0478)	0.137*** (0.0491)	0.112
Child's age-squared	-0.00298***	-0.00311 ** *	-0.00310***	-0.00450*	-0.00525**	-0.00225	-0.00470***	-0.00484***	-0.00398
Child's gandar – 1 if	(0.000349) 0.00970	(0.000524)	(0.000360) 0.00937	(0.00245) 0.0206	(0.00262) 0.0165	0.00166	(0.00172)	(0.00177) 0.00591	(0.00262) 0.0198
male				0	200			-	
	(0.00890)	(0.0154)	(0.00922)	(0.0143)	(0.0152)	(0.0127)	(0.0133)	(0.0142)	(0.0152)
Disability = 1 if child is disabled	-0.133**	-0.0591	-0.173**	0	0	-0.179**	0	0	0
	(0.0672)	(0.110)	(0.0694)	()	(·)	(0.0726)	(·)	()	(·)
Total infants in household	-0.0154***	-0.0571**	-0.0167***	-0.00398	-0.0126	-0.0167**	-0.00356	-0.0111	-0.0149
	(0.00414)	(0.0272)	(0.00582)	(0.00704)	(0.0104)	(0.00728)	(0.00645)	(0.00999)	(0.0105)
									Continued

Rabia Arif and Azam Chaudhry

									Continues
		Age group 5-	-17		Age group 5-	11		Age group 12-	-17
Dependent variable:						IV reg			
No. of days spent in school last week	OLS	IV reg with district FE	IV reg without district FE	OLS	IV reg with district FE	<i>without</i> district FE	OLS	IV reg with district FE	IV reg without district FE
Total boys aged 5–9 in household	-0.00652	-0.0225*	-0.00785	-0.0146*	-0.0167*	-0.000436	-0.0156**	-0.0164**	-0.0170*
	(0.00513)	(0.0127)	(0.00548)	(0.00845)	(0.00893)	(0.00736)	(0.00769)	(0.00793)	(00600.0)
Total girls aged 5–9 in household	0.0000541	-0.00928	-0.00309	-0.00924	-0.00581	0.00174	-0.00578	-0.00269	-0.0129
	(0.00524)	(0.00977)	(0.00541)	(0.00873)	(0.00951)	(0.00732)	(0.00793)	(0.00869)	(0.00943)
Total boys aged 10–17 in household	-0.0141 ***	-0.00735	-0.0187***	00600.0-	-0.00212	-0.0211***	-0.00987*	-0.00432	-0.00869
	(0.00392)	(0.00723)	(0.00407)	(0.00621)	(0.00878)	(0.00538)	(0.00575)	(0.00806)	(0.00899)
Total girls aged 10–17 in household	-0.0195***	-0.00267	-0.0228***	-0.00989	-0.00221	-0.0288***	-0.0117**	-0.00539	-0.00541
	(0.00380)	(0.0121)	(0.00419)	(0.00610)	(0.00918)	(0.00532)	(0.00564)	(0.00850)	(0.00913)
Household head if govt. emplovee	-0.0499***	0.0827	-0.0680***	-0.0439**	-0.0109	-0.0785* **	-0.0389**	-0.0104	-0.0321
-	(0.0143)	(0.0868)	(0.0185)	(0.0207)	(0.0358)	(0.0232)	(0.0192)	(0.0344)	(0.0343)
Household head if	0.00845	0.0382	0.00461	0.0270	0.0320	-0.0156	0.0298	0.0340^{*}	0.0380^{*}
	(0.0131)	(0.0271)	(0.0140)	(0.0198)	(0.0210)	(0.0187)	(0.0183)	(0.0193)	(0.0211)
Household head if laborer	-0.0150	0.0235	-0.0229*	0.000841	0.00687	-0.0292*	0.00114	0.00700	-0.00980
	(0.0119)	(0.0301)	(0.0127)	(0.0185)	(0.0199)	(0.0167)	(0.0170)	(0.0184)	(0.0197)
Household head's	0.0416***	-0.00597	0.0551***	0.0252	0.0113	0.0682***	0.0231	0.0102	0.0212
equeation II primary	(0,0118)	(0.0349)	(0.0131)	(0.0185)	(0.0226)	(0.0165)	(0.0170)	(0.0217)	(0.0238)
Household head's	0.0550***	0.0266	0.0644***	0.0288	0.0258	0.0855***	0.0222	0.0198	0.0282
	(0.0130)	(0.0263)	(0.0138)	(0.0198)	(0.0207)	(0.0183)	(0.0183)	(0.0190)	(0.0213)
Household head's	0.0675***	0.0661***	0.0714***	0.0391**	0.0438**	0.0956***	0.0432***	0.0461***	0.0376**
education if secondary	(0.0118)	(0.0174)	(0.0121)	(0.0177)	(0.0188)	(0.0163)	(0.0164)	(0.0171)	(0.0184)
									Continued

38 The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

									Continues
		Age group 5-1	17		Age group 5-	11		Age group 12-	-17
Dependent variable:			IV reg			IV reg			IV reg
No. of days spent in		IV reg with	without		IV reg with	without		IV reg with	without
school last week	OLS	district FE	district FE	OLS	district FE	district FE	OLS	district FE	district FE
Household head's education if higher	0.0843***	0.0964***	0.0717***	0.0498**	0.0490**	0.0926***	0.0527**	0.0518**	0.0280
	(0.0148)	(0.0232)	(0.0151)	(0.0223)	(0.0231)	(0.0203)	(0.0207)	(0.0212)	(0.0232)
Female bias	-0.00766	0.118	-0.0165	-0.0172	0.00937	-0.0379	-0.0402	-0.0172	0.00217
	(0.0235)	(0.0869)	(0.0271)	(0.0395)	(0.0469)	(0.0346)	(0.0371)	(0.0444)	(0.0474)
Wealth index 2	0.0366***	-0.0220	0.0529^{***}	0.0525**	0.0383	0.0479**	0.0597***	0.0492^{**}	0.0525*
	(0.0141)	(0.0427)	(0.0154)	(0.0245)	(0.0282)	(0.0188)	(0.0223)	(0.0251)	(0.0284)
Wealth index 3	0.0746***	-0.0588	0.101***	0.0641^{**}	0.0336	0.112***	0.0677***	0.0424	0.0619
	(0.0148)	(0.0875)	(0.0196)	(0.0249)	(0.0369)	(0.0234)	(0.0227)	(0.0342)	(0.0377)
Wealth index 4	0.106^{***}	-0.214	0.167^{***}	0.0872***	0.0169	0.202***	0.0921***	0.0344	0.0747
	(0.0162)	(0.205)	(0.0332)	(0.0267)	(0.0670)	(0.0387)	(0.0244)	(0.0624)	(0.0671)
Wealth index 5	0.0874***	-0.391	0.162^{***}	0.0276	-0.0777	0.234***	0.0437	-0.0455	0.00787
	(0.0192)	(0.305)	(0.0448)	(0.0307)	(0.0968)	(0.0511)	(0.0282)	(0.0929)	(0.0928)
Constant	5.077***	2.888^{**}	5.352^{***}	4.831***	4.151^{***}	5.615^{***}	4.789***	4.302^{***}	4.671***
	(0.0721)	(1.394)	(0.208)	(0.501)	(0.786)	(0.255)	(0.337)	(0.593)	(0.769)
No. of observations	60,900	60,900	006'09	22,833	22,833	38,067	26,688	26,688	22,833
R-squared	0.066		0.008	0.060			0.060	0.011	
Adjusted R-squared	0.065		0.007	0.057			0.058	0.009	
Notes: $* = p < 0.16$	$h_{\rm w}^{**} = n < 0$.	05. *** = p <	0.01. Standard	errors are give	n in parenthes	es below estim	nates. The distri	ct dummies. ur	ban dummv. and

lotes: $* = p < 0.10$, $*^* = p < 0.05$, $*^* = p < 0.01$. Standard errors are given in parentheses below estimates. The district dummies, urban dummy, and
ie variables for the distance to public and private schools were also added to the model. The coefficients' signs are consistent with the literature.
ource: Authors' calculations.

Overall, the last set of results implies that children in households that include an external migrant do not spend extra time on domestic chores. Hence, the effect of external migration on children's schooling outcomes is positive. These results are consistent with the literature, which finds that external migration has a positive, significant impact on school enrolments and accumulated level of schooling. Eased income constraints help households invest more in their children's schooling, and the number of days that children attended school last week was not affected significantly negatively.

7. Robustness Checks

In order to verify the result that external migration has a positive effect on children's schooling outcomes, we also conduct a series of robustness checks by testing dropouts and the number of hours spent on household chores as dependent variables. Both these variables may measure the negative effect of external migration on children.

7.1. Estimating the Impact of External Migration on Dropouts

Equation (13) estimates whether a child belonging to a migrant's household drops out of school the following year. This analysis was conducted for children aged 5–17 to determine the overall effect of migration on dropout rates.

Dropouts
$$_{ghi} = \beta_1 \overleftarrow{M}_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{hi} + \mu_{ghi}$$
 (13)

The dropouts variable takes the value of 1 if child *i* leaves school in 2008/09 assuming that he/she was previously enrolled. C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, B_{hi} is the child's gender, and \widetilde{M}_{ghi} represents the fitted values of migration from the first-stage regression.

Table 5 presents the results of our estimation for children dropping out of school if they belong to a household with an external migrant. The results show that the presence of an external migrant had no significant impact on dropouts among children aged 5–17 and 12–17. Another point worth noting was that, if district fixed effects are eliminated from

the regression equations, then the results in the table show that external migration has a significant negative effect on children aged 5–17. This analysis also holds true for the 12–17 age group.

Another important—and significant—variable is the proportion of women working outside the house, because it implies that households with a higher proportion of working women may create disruptions in children's schooling life if the latter are burdened with greater household responsibilities.

		Age gro	oup 5–17			Age gro	up 12–17	
				IV probit				IV probit
Dependent variable:			V probit <i>with</i>	without		-	IV probit with	without
Dropouts	OLS	IV reg	district FE	district FE	OLS	IV reg	district FE	district FE
External migrant in household	-0.207	0.0247	2.220	-3.240*	-0.170	0.00879	0.769	-4.217***
	(0.153)	(0.113)	(8.627)	(1.677)	(0.160)	(0.0819)	(4.097)	(1.483)
Household head's age	0.00856	0.000206	0.0164	-0.00310	0.0126	0.000273	0.0167	-0.00872
Household head's age-	(0.0117) -0.000127	(0.000419) -0.00000261	(0.0265) -0.000218	(0.0125) 0.0000190	(0.0133) -0.000185	(0.000436) -0.00000383	(0.0212) -0.000228	(0.0144) 0.0000652
squared								
	(0.000115)	(0.00000490)	(0.000291)	(0.000138)	(0.000133)	(0.00000452)	(0.000216)	(0.000163)
Household head's gender = 1 if male	0.0718	0.00845	0.776	-0.844	0.0883	0.00515	0.345	-1.093**
	(0.132)	(0.0328)	(2.454)	(0.551)	(0.144)	(0.0226)	(1.122)	(0.487)
Proportion of working women	0.135***	0.00309*	0.154***	0.108	0.130***	0.00439***	0.141***	0.0593
	(0.0350)	(0.00166)	(0.0352)	(0.0668)	(0.0394)	(0.00153)	(0.0541)	(0.0803)
Child's age	0.176***	-0.00134***	0.156	0.141^{**}	-0.102	-0.00682	-0.0823	-0.131
	(0.0592)	(0.000494)	(0.147)	(0.0573)	(0.286)	(0.00624)	(0.297)	(0.193)
Child's age-squared	-0.00240	0.000116***	-0.00211	-0.00185	0.00657	0.000304	0.00584	0.00657
	(0.00235)	(0.0000217)	(0.00288)	(0.00195)	(0.00984)	(0.000217)	(0.0103)	(0.00674)
Child's gender = 1 if male	0.0468	0.000608	0.0390	0.0443	0.0127	0.000341	0.0113	0.0192
	(0.0522)	(0.000586)	(0.0655)	(0.0432)	(0.0609)	(0.00127)	(0.0605)	(0.0409)
Disability = 1 if child is disabled		-0.00103						
		(0.00478)						
Total infants in household	-0.00838 (0.0248)	-0.000283 (0.000889)	-0.0270 (0.0663)	0.0256 (0.0245)	0.0157 (0.0279)	0.000304 (0.000822)	0.00864 (0.0421)	0.0441** (0.0192)
								Continued

Table 5: Estimating the effect of external migration on dropouts

42

								Continues
		Age gro	oup 5–17			Age gro	up 12–17	
Denendent variable:			IV probit with district	IV probit without			IV nrohit with	IV probit without
Dropouts	OLS	IV reg	FE	district FE	OLS	IV reg	district FE	district FE
Total boys aged 5–9 in household	0.00679	-0.0000781	-0.00302	0.0196	-0.0206	-0.000471	-0.0232	0.00211
	(0.0297)	(0.000532)	(0.0459)	(0.0247)	(0.0358)	(0.000794)	(0.0367)	(0.0265)
Total girls aged 5–9 in household	0.0316	0.000417	0.0243	0.0304	0.0565*	0.00140*	0.0571*	0.0303
	(0.0298)	(0.000382)	(0.0463)	(0.0249)	(0.0342)	(0.000782)	(0.0338)	(0.0297)
Total boys aged 10–17 in household	-0.0670***	-0.000633**	-0.0591	-0.0560***	-0.0737***	-0.00144**	-0.0696**	-0.0665***
	(0.0236)	(0.000271)	(0.0579)	(0.0217)	(0.0270)	(0.000635)	(0.0350)	(0.0234)
Total girls aged 10–17 in household	-0.0406*	-0.000375	-0.0317	-0.0404**	-0.0626**	-0.00119*	-0.0578	-0.0601***
	(0.0230)	(0.000342)	(0.0514)	(0.0192)	(0.0269)	(0.000647)	(0.0364)	(0.0215)
Household head if govt. emplovee	-0.0983	-0.000692	-0.0337	-0.154**	-0.106	-0.00224	-0.0819	-0.172***
	(0.0838)	(0.00271)	(0.276)	(0.0704)	(0.0918)	(0.00279)	(0.146)	(0.0615)
Household head if private employee	-0.305***	-0.00303***	-0.257	-0.270***	-0.302***	-0.00616***	-0.292***	-0.224**
	(0.0852)	(0.00111)	(0.298)	(0.0822)	(0:0030)	(0.00186)	(0.112)	(9660.0)
Household head if laborer	-0.159**	-0.00168	-0.124	-0.156***	-0.180**	-0.00398**	-0.171**	-0.143**
	(0.0646)	(0.00110)	(0.193)	(0.0556)	(0.0731)	(0.00167)	(0.0867)	(0.0671)
Household head's education if primary	0.0844	0.000977	0.0581	0.0973*	0.107	0.00259	0.0982	0.106**
-	(0.0611)	(0.00101)	(0.137)	(0.0500)	(0.0681)	(0.00166)	(0.0806)	(0.0506)
Household head's education if middle	-0.0591	-0.000767	-0.0612	-0.0364	-0.0871	-0.00181	-0.0873	-0.0457
	(0.0757)	(0.000931)	(0.0702)	(0.0666)	(0.0853)	(0.00175)	(0.0842)	(0.0678)
								Continued

IV probit <	rrobit IV probit district without EE district FE 0701 -0.00706 -0.			
Dependent variable: OLS IV reg FE district FE OLS Propouts OL IV reg FE district FE OL3 Household head's 0.000454 -0.0000122 0.000701 -0.00706 -0.0173 Household head's 0.109** -0.000135 -0.190** -0.0174 -0.0744 Household head's 0.1010 0.00059 (0.0668) (0.00744) -0.0744 Household head's -0.199** -0.00135 -0.168 -0.190** -0.254** Household head's -0.101 0.00159 (0.143) -0.0327 0.0743 Household head's 0.101 0.00159 0.143 0.0327 0.00733 Female bias 0.101 0.00156 (0.172) (0.112) 0.0171 Wealth index 2 0.00160 0.0132 0.01327 0.0373 0.00545 Wealth index 4 0.08030 (0.00160) (0.0740) (0.0157 0.0157 Wealth index 5 0.08030 (0.00258) (0.173) <th>district <i>without</i> FE district FE 0701 -0.00706 -0. 604) (0.0545) (0.</th> <th></th> <th></th> <th>IV probit</th>	district <i>without</i> FE district FE 0701 -0.00706 -0. 604) (0.0545) (0.			IV probit
Uropouts Unserved	re alstrict re 0701 -0.00706 -0. 504) (0.0545) (0.		IV probit with	without
Household head's 0.000454 -0.000701 -0.00706 -0.0173 education if secondary (0.0668) (0.000759) (0.0604) (0.0545) (0.0744) Household head's -0.199^{**} -0.00135 -0.168 -0.190^{**} -0.254^{**} Household head's -0.10035 -0.168 -0.190^{**} -0.254^{**} education if higher (0.102) (0.00106) (0.209) (0.0912) (0.115) Female bias 0.1011 0.00159 0.143 0.0327 0.00733 Wealth index 2 0.000166 -0.00106 0.1120 (0.1141) Wealth index 3 $0.00771)$ (0.00256) (0.172) (0.112) Wealth index 4 $0.00771)$ (0.00256) (0.173) (0.0974) Wealth index 5 -0.00711 $-0.00295)$ (0.173) (0.0974) Wealth index 4 $0.0803)$ (0.00255) (0.173) (0.0976) Wealth index 5 -0.00711 -0.02211 $0.0740)$ (0.0956) Wealth index 5 -0.00732 (0.0740) (0.0740) (0.0956) Wealth index 5 -0.00732 (0.00256) (0.173) (0.0740) (0.0976) Wealth index 5 (0.0890) (0.0740) (0.0740) (0.07574) Wealth index 5 (0.0800) (0.00758) (0.173) (0.0740) Wealth index 5 (0.0800) (0.00758) (0.173) (0.1056) Wealth index 6 (0.0800) (0.00770) (0.126) <tr< th=""><th>0701 -0.00706 -0.</th><th>ULS IV reg</th><th>district FE</th><th>district FE</th></tr<>	0701 -0.00706 -0.	ULS IV reg	district FE	district FE
contaction if higher (0.0668) (0.00759) (0.0604) (0.0545) (0.0745) Household head's -0.199^{**} -0.00135 -0.168 -0.190^{**} -0.254^{**} education if higher (0.102) (0.0016) (0.209) (0.0912) (0.115) female bias 0.101 0.00159 0.143 0.0327 0.0793 0.101 0.00159 0.143 0.0327 0.0793 Wealth index 2 0.00869 -0.00106 0.1120 (0.141) Wealth index 3 0.07710 (0.00160) (0.0829) (0.0740) (0.0934) Wealth index 4 0.00732 0.0141 0.0173 (0.0740) (0.0974) Wealth index 5 0.00730 (0.007200) (0.0740) (0.0974) (0.0740) Wealth index 5 0.00731 0.00732 (0.173) (0.0740) (0.0934) Wealth index 5 0.00730 (0.00730) (0.00730) (0.0740) (0.0934) Wealth index 5 0.00730 (0.00730) (0.0721) 0.0740 (0.0939) Wealth index 5 0.00730 (0.00730) (0.00730) (0.0740) (0.0740) Wealth index 5 0.00730 (0.00730) (0.00730) (0.00730) (0.0730) Wealth index 5 0.00730 (0.00730) (0.00730) (0.126) Wealth index 5 0.00730 (0.00730) (0.00730) (0.1350) Wealth index 6 0.00730 (0.00730) (0.00730) (0.126) </td <td>04) (0.0545) (0.</td> <td>0173 -0.000291</td> <td>-0.0127</td> <td>-0.0275</td>	04) (0.0545) (0.	0173 -0.000291	-0.0127	-0.0275
Household head's -0.00135 -0.068 -0.190^{**} -0.254^{**} education if higher 0.102 0.00159 0.0327 0.0327 0.254^{**} female bias 0.101 0.00159 0.143 0.0327 0.0793 Wealth index 2 0.00159 0.143 0.0373 0.0793 Wealth index 2 0.000159 0.1122 0.1141 0.0793 Wealth index 3 0.0771 0.00156 0.1722 0.1411 Wealth index 3 0.007106 -0.1099 0.0373 0.00545 Wealth index 4 0.00731 0.007295 0.1731 0.00793 Wealth index 5 0.00731 0.00211 0.2211 0.0157 Wealth index 5 0.000731 0.00731 0.00740 0.00999 Wealth index 5 0.000711 0.02211 0.1731 0.00574 Wealth index 5 0.000731 0.02211 0.1731 0.00999 Wealth index 5 0.000731 0.02211 0.1731 0.00994 0.126 Wealth index 5 0.000731 0.00977 0.0994 0.126 Wo of observations $64,897$ $69,081$ 0.0991 0.0994 0.128 No. of observations $64,897$ $69,081$ 2.0073 0.128		0.00163	(0.0763)	(0 0499)
education if highereducation if higher (0.102) (0.00106) (0.209) (0.0912) (0.115) Female bias 0.101 0.00159 0.143 0.0327 0.0793 Wealth index 2 0.00869 -0.00106 $0.1372)$ (0.141) Wealth index 3 $0.0771)$ (0.00166) 0.0373 0.00545 Wealth index 3 -0.00373 -0.00373 -0.00545 Wealth index 4 0.00730 (0.0720) (0.141) Wealth index 5 -0.00132 -0.1400 0.0141 0.0157 Wealth index 5 -0.00339 -0.00211 -0.2211 0.1736 (0.00574) Wealth index 5 $0.00258)$ (0.173) (0.0899) (0.00574) Wealth index 5 -0.00211 -0.2211 0.1366 0.00574 Wealth index 5 0.00271 0.02711 0.2211 0.1356 (0.126) Wealth index 5 0.00739 (0.00478) 0.04580 (0.126) (0.126) Wo of observations $64,897$ $69,081$ (0.491) (0.4921) (0.203) (0.126) No. of observations $64,897$ $69,081$ 2.0075 (0.128) (0.128)	58 -0.190** -0.	254** -0.00360*	-0.248**	-0.178
Female bias (0.102) (0.00106) (0.209) (0.0912) (0.115) Female bias 0.101 0.00159 0.143 0.0327 0.0793 Wealth index 2 0.01706 0.0152 0.1411 0.00745 0.1411 Wealth index 2 0.00771 0.00256 0.1122 0.01411 0.00745 Wealth index 3 0.00771 0.00706 0.0172 0.0141 0.00740 Wealth index 3 0.00732 0.0141 0.00740 0.00740 0.00740 Wealth index 4 0.0803 0.00711 0.0221 0.0177 0.00574 Wealth index 5 0.00211 0.2211 0.1730 0.09996 0.00574 Wealth index 5 0.00211 0.02211 0.1730 0.09996 0.00574 Wealth index 5 0.002718 0.02468 0.04668 0.00774 0.1025 Wealth index 5 0.002718 0.02711 0.2211 0.135 0.1266 W				
Female bias 0.101 0.00159 0.143 0.0327 0.0793 Wealth index 2 0.0177 (0.0256) (0.152) (0.141) Wealth index 2 -0.0869 -0.00166 -0.109 -0.0373 -0.0545 Wealth index 3 (0.0771) (0.00160) (0.0829) (0.0740) (0.0934) Wealth index 3 -0.0860 -0.00132 -0.1400 0.0141 0.0157 Wealth index 4 0.0839 (0.00255) (0.173) (0.0899) (0.0956) Wealth index 5 -0.00339 -0.00211 -0.2211 0.136 0.00574 Wealth index 5 -0.00211 -0.2211 0.136 0.00574 Wealth index 5 -0.00271 0.02711 0.0294 -0.126 Wealth index 5 -0.00478 -0.460 0.0994 -0.126 Wo of observations $64,897$ (0.458) (1.546) (0.128) No. of observations $64,897$ $69,081$ $28,075$	0) (0.0912) (0.	(115) (0.00207)	(0.121)	(0.113)
(0.117) (0.00256) (0.152) (0.112) (0.141) Wealth index 2 -0.0869 -0.00106 -0.0373 -0.00545 Wealth index 3 -0.0771) (0.0771) (0.0740) (0.0373) -0.00343 Wealth index 3 -0.00132 -0.109 (0.0740) (0.0934) Wealth index 4 -0.0860 -0.00132 -0.1400 0.0141 0.0157 Wealth index 4 -0.0839 (0.00251) (0.173) (0.0899) (0.0574) Wealth index 5 -0.00211 -0.2211 0.1356 (0.0574) (0.0574) Wealth index 5 -0.00211 -0.2211 0.1356 (0.1057) (0.1057) Wealth index 5 -0.00274 (0.0458) (0.458) (0.126) (0.126) Wealth index 5 -0.000774 (0.0774) (0.2033) (0.126) Wealth index 5 -0.000774 (0.00774) (0.128) (0.484) (0.484) (0.128) Constant $-3.960***$.3 0.0327 0.0	793 0.00246	0.0954	-0.0208
Wealth index 2 -0.0869 -0.00106 -0.109 -0.0373 -0.00545 Wealth index 3 0.0771 (0.0771) (0.0740) (0.0733) -0.00334 Wealth index 3 -0.0860 -0.00132 -0.140 (0.0740) (0.0934) Wealth index 4 0.0839 (0.00150) (0.0829) (0.0157) (0.0956) Wealth index 4 -0.0839 -0.00211 -0.221 0.0137 (0.0956) Wealth index 5 -0.00211 -0.221 0.136 0.00574 Wealth index 5 -0.00211 -0.221 0.135 (0.105) Wealth index 5 $-0.271**$ -0.00478 0.460 0.0394 -0.126 Wealth index 5 $-0.271**$ -0.00478 -0.460 0.0394 -0.126 Wealth index 5 $-0.271**$ -0.00478 -0.460 0.0231 0.128 Wealth index 5 -0.00436 $-4.434***$ -2.404 -1.868 No. of observations $64,897$	52) (0.112) (0.	(0.00388) (0.00388)	(0.154)	(0.113)
Wealth index 3 (0.0771) (0.00160) (0.0829) (0.0740) (0.0934) Wealth index 3 -0.0860 -0.00132 -0.140 0.0141 0.0157 Wealth index 4 (0.0839) (0.0295) (0.173) (0.0956) Wealth index 4 -0.0839 -0.00211 -0.221 0.0137 $0.0956)$ Wealth index 5 $-0.0890)$ (0.00558) (0.458) (0.135) (0.00574) Wealth index 5 $-0.271**$ -0.00478 -0.460 0.0994 -0.126 Wealth index 5 $-0.271**$ -0.00478 -0.460 0.0994 -0.126 Wealth index 5 $-0.271**$ -0.00478 -0.460 0.0231 0.128 Wealth index 5 $-0.271**$ -0.00478 -0.460 0.0231 0.128 Wealth index 5 $-0.271**$ -0.00478 -0.460 0.233 0.128 No of observations $64,897$ $69,081$ 2.068 2.063 2.063	-0.0373 -0.0	00545 -0.000389	-0.0198	0.0543
Wealth index 3 -0.0860 -0.00132 -0.140 0.0141 0.0157 Wealth index 4 0.0803 (0.0295) (0.173) (0.0899) (0.0956) Wealth index 4 -0.0839 -0.00211 -0.221 0.136 0.00574 Wealth index 5 $-0.0890)$ (0.0658) (0.458) (0.135) (0.105) Wealth index 5 -0.271^{**} -0.00478 -0.460 0.0994 -0.126 Wealth index 5 -0.271^{**} -0.00478 -0.460 0.0994 -0.126 Wealth index 6 -0.271^{**} -0.00478 -0.440 (0.128) Wo of observations (0.484) (0.0404) (0.458) (1.546) (2.088) No. of observations $64,897$ $69,081$ $28,075$	829) (0.0740) (0.	0.00238) (0.00238)	(0.111)	(0.0680)
(0.0803) (0.00295) (0.173) (0.0899) (0.0956) Wealth index 4 -0.0839 -0.00211 -0.221 0.136 0.00574 Wealth index 5 -0.0890 (0.0658) (0.458) (0.135) (0.105) Wealth index 5 -0.271^{**} -0.00478 -0.460 0.0994 -0.126 Wealth index 5 -0.271^{**} -0.00478 -0.460 0.0994 -0.126 Wealth index 5 -0.271^{**} -0.00478 -0.460 0.0994 -0.126 Wealth index 5 -0.271^{**} -0.00478 -0.440 0.128 0.128 No of observations $64,897$ $69,081$ 2.008 2.008 0.075	40 0.0141 0.0	157 0.0000477	-0.0117	0.137^{*}
Wealth index 4 -0.0839 -0.00211 -0.221 0.136 0.00574 Wealth index 5 0.0890) (0.0658) (0.458) (0.135) (0.105) Wealth index 5 -0.271** -0.00478 -0.460 0.0994 -0.126 Wealth index 5 -0.271** -0.00478 -0.460 0.0994 -0.126 Constant -3.960*** -0.000436 -4.434*** -2.404 -1.868 No. of observations 64,897 69,081 28,075	73) (0.0899) (0.)956) (0.00313)	(0.153)	(0.0750)
(0.0890) (0.00658) (0.458) (0.135) (0.105) Wealth index 5 -0.271** -0.00478 -0.460 0.0994 -0.126 Wealth index 5 (0.112) (0.00977) (0.571) (0.203) (0.128) Constant -3.960*** -0.000436 -4.434*** -2.404 -1.868 No. of observations 64,897 69,286 64,897 69,081 28,075	21 0.136 0.0	0574 -0.000608	-0.0520	0.269^{**}
Wealth index 5 -0.271** -0.00478 -0.460 0.0994 -0.126 Wealth index 5 -0.112) (0.00977) (0.571) (0.203) (0.128) Constant -3.960*** -0.000436 -4.434*** -2.404 -1.868 No. of observations (0.484) (0.0404) (0.458) (1.546) (2.088)	58) (0.135) (0.	(0.00532) (0.00532)	(0.272)	(0.110)
(0.112) (0.00977) (0.571) (0.203) (0.128) Constant -3.960*** -0.000436 -4.434*** -2.404 -1.868 (0.484) (0.0404) (0.458) (1.546) (2.088) No. of observations 64,897 69,286 64,897 69,081 28,075	60 0.0994 -0.	126 -0.00349	-0.210	0.309*
Constant -3.960*** -0.000436 -4.434*** -2.404 -1.868 (0.484) (0.0404) (0.458) (1.546) (2.088) No. of observations 64,897 69,286 64,897 69,081 28,075	71) (0.203) (0.	(0.00767) (0.00767)	(0.380)	(0.179)
(0.484) (0.0404) (0.458) (1.546) (2.088) No. of observations 64,897 69,286 64,897 69,081 28,075	34*** -2.404 -1.	368 0.0523	-2.324	0.516
No. of observations 64,897 69,286 64,897 69,081 28,075	58) (1.546) (2.)	0.0593) (0.0593)	(2.775)	(2.052)
	.97 69,081 28,	075 29,857	28,075	29,857
R-squared 0.002		0.007		
Adjusted R-squared 0.001		0.005		
Pseudo-R-squared 0.125				

פ , **Notes:** f = p < 0.10, f = p < 0.00, f = p < 0.01. Standard errors are given in parentheses below estimates. The district dummies, urban the variables for the distance to public and private schools were also added to the model. The coefficients' signs are consistent with the literature. **Source:** Authors' calculations.

7.2. Estimating the Impact of External Migration on the Number of Hours Spent on Household Chores

The number of hours in a day that the child spends on household chores is estimated by the following equation:

No. of hours spent on household chores ghi

$$=\beta_1 \widetilde{M}_{ghi} + \beta_2 C_{ghi} + \beta_3 X_{hi} + \beta_4 B_{hi} + \mu_{ghi}$$
(14)

where S_{ghi} is the schooling outcome, C_{ghi} is a vector of the child's characteristics, X_{hi} is a vector of the household's characteristics, B_{hi} is the child's gender, and \widetilde{M}_{ghi} represents the fitted values of migration from the first-stage regression.

Table 6 shows that external migration does not have a significant effect on the number of hours children spend on household chores. If, however, we eliminate district fixed effects from the regression, then external migration appears to have a significant negative effect on the number of hours spent by children on household chores. This implies that, since households' resources increase through remittances, they are able to engage extra assistance. Hence, the number of hours that children spend on household chores decreases significantly.

The results for the subsample of younger children support the result that children face no added responsibility when there is an external migrant in their household and, hence, no significant effect of external migration on the hours they spend on household chores. Other variables added to the specification reveal that male children spend significantly less time on household chores than female children. Likewise, children are likely to spend significantly more time on household chores as the proportion of working women in their household increases. It is also interesting to note that, for children who belong to wealthier families—those ranked higher on the wealth indices score—the number of hours spent on household chores drops significantly than if they had been ranked within the first wealth index.

The results for the subsample of older children show that the impact of external migration on the number of hours spent on household chores

remains insignificant. However, the coefficient of the external migrant variable becomes significant and negative as soon as the district fixed effects are eliminated from the regression.

In all, these results and robustness checks strengthen our results indicating that children from households with external migrants are better off than those from households without external migrants because of the greater availability of resources that remittances yield. Therefore, we can argue that external migration has a positive influence on children's school enrolments and accumulated level of schooling, and also helps reduce their household workload.

		Age group 5-	-17		Age group 5-	11	1	Age group 12-	17
Dependent variable: No. of hours spent on		IV probit <i>with</i> district	IV probit without		IV probit <i>with</i> district	IV probit without		IV probit with district	IV probit without
household chores	OLS	FE	district FE	OLS	E	district FE	OLS	FE	district FE
External migrant in household	-0.233	-50.80	-9.929**	-0.590***	-0.933	-3.509	0.426	5.955	-34.57*
	(0.200)	(109.9)	(4.049)	(0.227)	(6.724)	(3.373)	(0.375)	(9.108)	(18.84)
Household head's age	-0.0320*	-0.163	-0.0251	-0.0341*	-0.0349	-0.00832	-0.0328	-0.0151	-0.116
	(0.0170)	(0.286)	(0.0214)	(0.0193)	(0.0249)	(0.0218)	(0.0326)	(0.0440)	(0.0790)
Household head's age- squared	0.000318*	0.00192	0.000329	0.000367**	0.000377	0.000143	0.000268	0.0000717	0.00129
	(0.000164)	(0.00349)	(0.000222)	(0.000186)	(0.000276)	(0.000221)	(0.000313)	(0.000452)	(0.000847)
Household head's gender = 1 if male	-0.0979	-12.93	-2.614**	0.169	0.0807	-0.580	-0.378	1.000	-9.481*
	(0.247)	(27.90)	(1.104)	(0.305)	(1.759)	(0.952)	(0.419)	(2.307)	(5.015)
Proportion of working	0.199***	-0.328	0.356^{***}	0.238***	0.234**	0.392***	0.148	0.189	0.246
women								1	
	(0.0707)	(1.152)	(0.0865)	(0.0818)	(0.118)	(0.0947)	(0.129)	(0.147)	(0.229)
Child's age	0.171*	0.432	0.221^{*}	-0.00262	-0.00224	-0.0422	-0.0154	0.680	-5.200
Child's age-sourced	(0.103) 0.0208***	0.596)	(0.114) 0.0192***	(0.209) 0.0315**	(0.210) 0.0315**	(0.222) 0.0350***	(3.649) 0.0325	(3.861) 0.00600	(5.702) 0.230
and adams	(0.00517)	(0.0251)	(0.00568)	(0.0127)	(0.0127)	(0.0134)	(0.141)	(0.149)	(0.219)
Child's gender = 1 if male	-0.543***	-0.371	-0.591***	-0.260***	-0.259***	-0.327***	-1.065***	-1.107***	-0.872***
	(0.0759)	(0.399)	(0.0838)	(0.0875)	(0.0893)	(0.0929)	(0.154)	(0.170)	(0.271)
Disability = 1 if child is disabled	-0.338	-1.446	-0.406	-0.258	-0.265	-0.243	0	0	0
	(0.613)	(2.644)	(0.670)	(0.550)	(0.568)	(0.586)	(:)	()	()
Total infants in household	-0.00135 (0.0348)	0.383 (0.838)	0.0728 (0.0506)	0.00507 (0.0377)	0.00807 (0.0697)	0.0364 (0.0505)	0.00415 (0.0726)	-0.0271 (0.0896)	0.199 (0.164)

Continued...

									Continues
		Age group 5-	-17		Age group 5-	-11		Age group 12-	-17
Dependent variable: No. of hours spent on household chores	510	IV probit <i>with</i> district FF	IV probit without district FF	SIC	IV probit <i>with</i> district FF	IV probit <i>without</i> district FF	510	IV probit <i>with</i> district FF	IV probit <i>without</i> district FF
Total boys aged 5–9 in household	0.232***	0.386	0.201***	0.130***	0.131**	0.0834	0.316***	0.295***	0.368***
	(0.0425)	(0.342)	(0.0477)	(0.0480)	(0.0532)	(0.0519)	(0.0856)	(0.0931)	(0.137)
Total girls aged 5–9 in household	-0.0171	-0.0875	-0.108**	-0.0694	-0.0694	-0.137***	0.192**	0.218**	-0.0527
	(0.0440)	(0.172)	(0.0482)	(0.0497)	(0.0497)	(0.0525)	(0.0887)	(0660.0)	(0.148)
Total boys aged 10–17 in household	-0.136***	-0.194	-0.205***	-0.0941**	-0.0938**	-0.148***	-0.157**	-0.126	-0.420***
	(0.0338)	(0.141)	(0.0370)	(0.0382)	(0.0385)	(0.0402)	(0.0674)	(0.0854)	(0.147)
Total girls aged 10–17 in household	-0.0656**	-0.177	-0.159***	-0.0917**	-0.0922**	-0.162***	-0.114*	-0.0920	-0.344***
	(0.0332)	(0.248)	(0.0373)	(0.0376)	(0.0390)	(0.0401)	(0.0658)	(0.0758)	(0.122)
Household head if govt. employee	0.131	-1.337	-0.127	0.211	0.201	0.0964	0.0261	0.176	-0.826
	(0.132)	(3.200)	(0.184)	(0.152)	(0.258)	(0.190)	(0.242)	(0.347)	(0.612)
Household head if private employee	0.0377	-0.631	-0.00116	0.148	0.143	0.177	-0.150	-0.0790	-0.519
	(0.117)	(1.468)	(0.143)	(0.133)	(0.162)	(0.151)	(0.222)	(0.253)	(0.424)
Household head if laborer	-0.0971	-0.567	-0.0620	-0.144	-0.147	-0.0641	-0.0334	0.0118	-0.163
	(0.100)	(1.038)	(0.118)	(0.113)	(0.132)	(0.126)	(0.193)	(0.209)	(0.319)
Household head's	0.132	0.560	0.264^{**}	0.222**	0.225^{*}	0.262**	-0.0306	-0.0866	0.492
	(0.0982)	(0.946)	(0.114)	(0.111)	(0.121)	(0.120)	(0.188)	(0.211)	(0.356)
Household head's education if middle	0.202*	0.290	0.140	0.280**	0.281**	0.158	0.0379	0.0587	0.0177
	(0.111)	(0.276)	(0.121)	(0.126)	(0.130)	(0.134)	(0.213)	(0.218)	(0.295)
									Continued

48 The Effects of External Migration on Enrolments, Accumulated Schooling, and Dropouts in Punjab

									Continues
		Age group 5-	-17		Age group 5-	-11		Age group 12-	-17
Dependent variable:		IV probit with district	IV probit		IV probit with district	IV probit		IV probit with district	IV probit
household chores	OLS	FE	district FE	OLS	FE	district FE	OLS	FE	district FE
Household head's	-0.0225	0.160	-0.0540	-0.000928	0.00124	-0.0432	-0.0906	-0.0870	-0.138
education if secondary									
	(0.103)	(0.438)	(0.113)	(0.118)	(0.125)	(0.126)	(0.195)	(0.197)	(0.268)
Household head's	-0.219	-0.163	-0.463***	-0.215	-0.214	-0.436***	-0.255	-0.262	-0.428
education if higher									
	(0.136)	(0.272)	(0.146)	(0.152)	(0.153)	(0.159)	(0.262)	(0.265)	(0.371)
Female bias	0.0403	1.123	0.152	0.224	0.233	0.206	-0.284	-0.303	-0.171
	(0.218)	(2.384)	(0.248)	(0.241)	(0.304)	(0.268)	(0.455)	(0.461)	(0.631)
Wealth index 2	-0.0895	0.632	0.268^{**}	-0.164	-0.160	0.0865	-0.0387	-0.117	0.753^{*}
	(0.109)	(1.580)	(0.131)	(0.119)	(0.152)	(0.133)	(0.223)	(0.260)	(0.426)
Wealth index 3	-0.688***	0.502	-0.151	-0.665***	-0.658***	-0.358**	-0.865***	-1.027***	0.670
	(0.118)	(2.596)	(0.160)	(0.130)	(0.188)	(0.152)	(0.238)	(0.359)	(0.665)
Wealth index 4	-0.964***	1.649	0.0459	-1.008***	-0.990***	-0.437*	-1.001***	-1.279**	1.513
	(0.134)	(5.686)	(0.257)	(0.150)	(0.384)	(0.238)	(0.264)	(0.530)	(1.036)
Wealth index 5	-1.065***	2.721	0.258	-1.091***	-1.066**	-0.388	-1.152***	-1.567**	2.378
	(0.165)	(8.234)	(0.340)	(0.185)	(0.536)	(0.304)	(0.319)	(0.756)	(1.472)
Constant	6.758***	20.49	7.542^{***}	6.748***	6.846^{***}	6.140^{***}	8.587	2.277	51.40
	(0.715)	(29.87)	(1.381)	(1.026)	(2.181)	(1.456)	(23.61)	(26.01)	(40.29)
No. of observations	29,222	29,222	29,222	18,672	18,672	18,672	10,550	10,550	10,550
R-squared	0.169		0.020	0.154	0.153	0.050	0.143	0.126	
Adjusted R-squared	0.167		0.019	0.151	0.151	0.049	0.138	0.120	
Notes: * = p < 0.10, **	= p < 0.05,	*** = p < 0.0	11. Standard err	ors are given ir	n parentheses	below estimate	ss. The district	dummies, urb	an dummy, and

the variables for the distance to government and private schools were also added to the model. The coefficients' signs are consistent with the literature. Source: Authors' calculations.

8. Conclusion

We find that external migration has a positive and significant impact on schooling attainment, which means that, in the case of Pakistan, the positive effect associated with remittances outweighs the negative effect of external migration due to the absenteeism of a family member. The results show that children from external migrant households not only have higher enrolment levels than those from nonmigrant households, but also show a higher level of accumulated schooling. On repeating the analysis with subsamples of older and younger children, we see that the positive effect of external migration on enrolment is larger for younger children than for older children. Mansuri's (2006) results for children in rural Pakistan indicate that the effect of external migration for older children is positive and significant. However, since the instrumental variables used by Mansuri (2006) are current migration community averages, the coefficients estimated were not fully corrected for bias since current migration rates and current enrolments might have simultaneously affected the unobserved variables.

The results also show that a child's accumulated level of schooling is positively influenced by external migration. The effect of migration on accumulated schooling is significantly positive only for older children. Mansuri's (2006) results reveal that the effect of external migration remains significantly positive for both younger and older children. Another point worth noting is that all the coefficients estimated in her study are smaller than those in ours, which shows that the impact of external migration on school enrolment and accumulated level of schooling is larger than that reported by previous studies.

Since school enrolments does not give us a complete picture of the impact of external migration on children's schooling outcomes, we have also tested other hypotheses such as its impact on the number of hours spent by children on household chores, the number of days spent last week in school, and the number of dropouts. Most previous studies do not have specific data to address these questions in detail. The results for dropouts—conditional on the child having been enrolled last year—indicated that external migration had no significant effect on it. The coefficient for external migration became significantly negative when district dummies were omitted from the regression. The results remained consistent with those from the regressions run on the major dataset for all children aged 5–17.

Similarly, the results for the number of days spent by a child in school and the number of hours spent on domestic chores indicate that there is no significant effect on the two dependent variables even if there is an external migrant in the household. The education of children in the migrant's household is not affected bv the added external responsibilities they might face because of an absent family member. Moreover, they do not have to spend significantly more hours on domestic chores, nor is the number of school days attended last week by the two different types of households mentioned in the study significantly different from one other.

The other variables added to the regression are reasonably consistent with other studies. The distance from the school significantly affects the accumulated level of schooling. The coefficient indicates that, as the distance increases, a child's accumulated schooling drops significantly. Likewise, if the household head has a higher level of education, this affects a child's schooling positively. Wealth indices also play a key role in determining school enrolment, accumulated levels of schooling, and the number of hours spent on household chores. Children from higherincome families have access to better resources and, hence, acquire higher levels of accumulated schooling than children from poor families. They also spend significantly less time on household chores than children from lower-income families. Our results also indicate the existence of a gender bias in rural Punjab: Boys have higher enrolment rates than girls but the latter's accumulated level of schooling is significantly higher than that of boys. Likewise, the number of hours that boys spend on household chores is significantly lower than that of girls.

As the proportion of working women in a household increases, the number of hours that a child spends on domestic chores increases significantly, as does the number of dropouts. This indicates that children face greater responsibilities when the proportion of women working outside the house rises; children may end up leaving school and spending more time on domestic chores. However, we also see that the accumulated level of schooling for children enrolled in schools increases significantly, which can be explained by the argument that households with a greater proportion of working women are more aware of the opportunities associated with higher education in the market. Our results are also consistent with the similar studies conducted for other countries. Hanson and Woodruff (2003), in their study on Mexico, conclude that children complete more years of schooling if they belong to households with external migrants. Edwards and Ureta (2003), in their study on El Salvador, find that the retention rate among children in households receiving remittances is lower than those without remittances. Alcaraz, Chiquiar, and Salcedo (2012) conduct a similar study for Mexico and conclude that the remittance crisis has had a negative impact on children's schooling and a positive impact on child labor.

The findings of this working paper imply that external migration has a significant positive effect on human capital formation in the Punjab, Pakistan. Based on the positive impact of external migration on human capital outlined, one can argue in favor of external migration as well as for improvements in formal channels to promote migration.

References

- Acosta, P. (2006). Labor supply, school attendance, and remittances from international migration: The case of El Salvador (Policy Research Working Paper No. 3903). Washington, DC: World Bank.
- Adams, R. H., Jr. (1998). Remittances, investment, and rural asset accumulation in Pakistan. *Economic Development and Cultural Change*, 47(1), 155–173. doi:10.1086/452390
- Adams, R. H., Jr. (2005). *Remittances, household expenditure and investment in Guatemala* (Policy Research Working Paper No. 3532). Washington, DC: World Bank.
- Alcaraz, C., Chiquiar, D., & Salcedo, A. (2012). Remittances, schooling, and child labor in Mexico. *Journal of Development Economics*, 97(1), 156–165. doi:10.1016/j.jdeveco.2010.11.004
- Amuedo-Dorantes, C., & Pozo, S. (2007). Remittances and the educational attainment of children: The Dominican case. Mimeo.
- Arif, G. M. (2009). Economic and social impacts of remittances on households: The case of Pakistani migrants working in Saudi Arabia. Geneva, Switzerland: International Organization for Migration.
- Bansak, C., & Chezum, B. (2009). How do remittances affect human capital formation of school-age boys and girls? *American Economic Review*, 99(2), 145–48. doi:10.1257/aer.99.2.145
- Becker, G. S. (1991). *A treatise on the family*. Cambridge, MA: Harvard University Press.
- Becker, G. S., & Tomes, N. (1976). Child endowments and the quantity and quality of children. *Journal of Political Economy*, 84(4), 143–162. doi:10.1086/260536
- Ben-Porath, Y. (1967). The production of human capital and the life cycle of earnings. *Journal of Political Economy*, 75(4, Pt. 1), 352–65. doi:10.1086/259291

- Conway, D., & Cohen, J. H. (1998). Consequences of migration and remittances for Mexican transnational communities. *Economic Geography*, 74(1), 26–44. doi:10.2307/144342
- Durand, J., Massey, D. S., & Zenteno, R. M. (2001). Mexican immigration in the United States. *Latin American Research Review*, 36(1), 107–127.
- Edwards, A. C., & Ureta, M. (2003). International migration, remittances, and schooling: Evidence from El Salvador. *Journal of Development Economics*, 72(2), 429–461. doi:10.1016/S0304-3878(03)00115-9
- Government of Pakistan, Bureau of Emigration and Overseas Employment. (2009). [The placement of Pakistani migrants abroad]. Unpublished data.
- Government of the Punjab, Bureau of Statistics. (2009). *Multiple indicator cluster survey (MICS): Punjab 2007–08*. Lahore, Pakistan: Author.
- Hanson, G. H., & Woodruff, C. (2003). *Emigration and educational attainment in Mexico*. Mimeo. Retrieved from http://irps.ucsd.edu/assets/022/8772.pdf
- Hoddinott, J. (1994). A model of migration and remittances applied to western Kenya. Oxford Economic Papers, 46(3), 459–476.
- Jacoby, H. G. (1994). Borrowing constraints and progress through school: Evidence from Peru. *Review of Economics and Statistics*, 76(1), 151–160. doi:10.2307/2109833
- Jacoby, H. G., & Skoufias, E. (1997). Risk, financial markets and human capital in a developing country. *Review of Economic Studies*, 64(3), 311–335. doi:10.2307/2971716
- Kugler, M. (2006). Migrant remittances, human capital formation and job creation externalities in Colombia (Borradores de Economía No. 2463). Bogotá, Colombia: Banco de la República.
- Levhari, D., & Weiss, Y. (1974). The effect of risk on the investment in human capital. *American Economic Review*, 64(6), 950–963.

- Lowell, B. L., & Findlay, A. (2001). *Migration of highly skilled persons* from developing countries: Impact and policy responses (International Migration Paper No. 44). Geneva, Switzerland: International Labour Office.
- Mansuri, G. (2006). *Migration, school attainment and child labor: Evidence from rural Pakistan* (Policy Research Working Paper No. 3945). Washington, DC: World Bank.
- Marcelli, E. A., & Cornelius, W. A. (2001). The changing profile of Mexican migrants to the United States: New evidence from California and Mexico. *Latin American Research Review*, *36*(3), 105–131.
- Martin, P. (1990) Labor migration and economic development. Washington, DC: Commission for the Study of International Migration and Cooperative Economic Development.
- Massey, D. S., & Basem, L. (1992). Determinants of savings remittances and spending patterns among Mexican migrants to the United States. *Sociological Inquiry*, 62, 186–207.
- Massey, D. S., Arango, J., Hugo, G., Kouaouci, A., Pellegrino, A., & Taylor, J. E. (1993). Theories of international migration: A review and appraisal. *Population and Development Review*, 19(3), 431–466. doi:10.2307/2938462
- McKenzie, D., & Rapoport, H. (2004). Network effects and the dynamics of migration and inequality: Theory and evidence from Mexico (Working Paper No. 063). Durham, NC: Bureau for Research and Economic Analysis of Development.
- Mincer, J. (1978). Family migration decisions. *Journal of Political Economy*, 86(5), 749–773. doi:10.1086/260710
- Rapoport, H., & Docquier, F. (2005). *The economics of migrants' remittances* (Discussion Paper No. 1531). Bonn, Germany: Institute for the Study of Labor.
- Russell, S. (1992). Migrant remittances and development. *International Migration: Quarterly Review, 30*(3/4), 267–287.

- Sawada, Y., & Lokshin, M. (2001). Household schooling decision in rural Pakistan (Policy Research Working Paper No. 2541). Washington, DC: World Bank.
- Stark, O., & Lucas, R. E. B. (1988). Migration, remittances, and the family. *Economic Development and Cultural Change*, 36(3), 465–481. doi:10.1086/451670
- Taylor, E. J., Arango, J., Hugo, G., Kouaouci, A., Massey, D. S., & Pellegrino, A. (1996a). International migration and community development. *Population Index*, 62(3), 397–418. doi:10.2307/3645924
- Taylor, E. J., Arango, J., Hugo, G., Kouaouci, A., Massey, D. S., & Pellegrino, A. (1996b). International migration and national development. *Population Index*, 62(2), 181–212. doi:10.2307/3646297
- Thieme, S., & Wyss, S. (2005). Migration patterns and remittance transfer in Nepal: A case study of Sainik Basti in western Nepal. *International Migration*, 43(5), 59–98. doi:10.1111/j.1468-2435.2005.00342.x
- United Nations Population Fund. (2009). *State of world population* 2009. New York, NY: Author.
- World Bank. (2005). *Tonga: Tonga education support project* (Report No. 32213). Washington, DC: Author.
- World Bank. (2006). Global economic prospects 2006: Economic implications of remittances and migration. Washington, DC: Author.
- Yang, D. (2006). Why do migrants return to poor countries? Evidence from Philippine migrants' responses to exchange rate shocks. *Review of Economics and Statistics*, 88(4), 715–735. doi:10.1162/rest.88.4.715
- Zeldes, S. P. (1989). Consumption and liquidity constraints: An empirical investigation. *Journal of Political Economy*, 97(2), 305–346. doi:10.1086/261605

Lahore School of Economics Centre for Research in Economics & Business

Working Paper Series

No. 02-12

The Determinants of Child Health and Nutritional Status in Punjab: An Economic Analysis Uzma Afzal

No. 01-12 Investigating the Proposed Changes to Pakistan's Corporate Bankruptcy Code Ali Hasanain, Syed Ahsan Ahmad Shah

No. 02-11

Cross-Country Growth Spillovers: Separating the Impact of Cultural Distance from Geographical Distance

Azam Chaudhry, Rabia Ikram

No. 01-11

The Determinants of Female Labor Force Participation in Pakistan: An Instrumental Variable Approach Mehak Ejaz

No. 01-10

The Determinants of Interest Rate Spreads in Pakistan's Commercial Banking Sector Ayesha Afzal, Nawazish Mirza

No. 03-09

Evaluating the Impact of Microcredit on Women's Empowerment in Pakistan Salman Asim

No. 02-09 **Speculative Bubbles in Karachi Stock Exchange** Nawazish Mirza

No. 01-09 Economic Development: A View From the Provinces Khalid Ikram

No. 02-08 **Agricultural Growth in Irrigated Punjab: Some Issues and Policies** Mahmood Hassan Khan

No. 01-08 **Size and Value Premium in Karachi Stock Exchange** Nawazish Mirza

Policy Paper Series

No. 01-11 **Pakistan and Lessons from East Asia: Growth, Equity, and Governance** Khalid Ikram

No. 01-10 A Strategy for Reversing Pakistan's Dismal Export Performance Hamna Ahmed, Mahreen Mahmud, Naved Hamid, Talal-Ur-Rahim

These papers can be accessed at: www.creb.org.pk

The Lahore School of Economics (established in 1993) is one of Pakistan's leading centres of learning for teaching and research in economics, finance and business administration. Its objectives are (i) to train young Pakistanis as professional economists, finance managers, accountants, financial analysts, bankers, and business executives, and (ii) to undertake research in economics, management, finance, and banking to deepen the understanding of major facts, issues, and policies.

The Centre for Research in Economics and Business (CREB) is an independent research centre at the Lahore School of Economics. CREB's mission is to conduct and facilitate research, coordinate and manage the Lahore School's postgraduate program, and promote discussion on policy issues facing Pakistan. The research focus at CREB is on the management of the Pakistan economy, income distribution and poverty, and the role of the modern services sector in the area of economics; and financial markets in the area of business management.

The Lahore School's publication program comprises the Lahore Journal of Economics, Lahore Journal of Policy Studies, Lahore Journal of Business, a Text Book Series, Lahore School Case Study Journal, the CREB Working Paper Series, and CREB Policy Paper Series. The program encourages both in-house and external contributors.

